REVIEW

Open Access

Review of potential medical treatments for middle ear cholesteatoma



Matthias Schürmann¹, Peter Goon^{1,2} and Holger Sudhoff^{1*}

Abstract

Middle ear cholesteatoma (MEC), is a destructive, and locally invasive lesion in the middle ear driven by inflammation with an annual incidence of 10 per 100,000. Surgical extraction/excision remains the only treatment strategy available and recurrence is high (up to 40%), therefore developing the first pharmaceutical treatments for MEC is desperately required. This review was targeted at connecting the dysregulated inflammatory network of MEC to pathogenesis and identification of pharmaceutical targets. We summarized the numerous basic research endeavors undertaken over the last 30+ years to identify the key targets in the dysregulated inflammatory pathways and judged the level of evidence for a given target if it was generated by in vitro, in vivo or clinical experiments. MEC pathogenesis was found to be connected to cytokines characteristic for Th1, Th17 and M1 cells. In addition, we found that the inflammation created damage associated molecular patterns (DAMPs), which further promoted inflammation. Similar positive feedback loops have already been described for other Th1/Th17 driven inflammatory diseases (arthritis, Crohn's disease or multiple sclerosis). A wide-ranging search for molecular targeted therapies (MTT) led to the discovery of over a hundred clinically approved drugs already applied in precision medicine. Based on exclusion criteria designed to enable fast translation as well as efficacy, we condensed the numerous MTTs down to 13 top drugs. The review should serve as groundwork for the primary goal, which is to provide potential pharmaceutical therapies to MEC patients for the first time in history.

Keywords: Cholesteatoma, Chronic inflammation, Innate immune system, Adaptive immune response, Positive feedback loops, Precision medicine

Plain English summary

Middle ear cholesteatoma (MEC) is a destructive and locally invasive ulcerated lesion in the middle ear driven by inflammation which occurs in 10 out of 100,000 people annually. Surgical extraction/excision is the only treatment strategy available and recurrence is high (up to 40% after ten years), therefore developing the first pharmaceutical treatments for MEC is desperately required. This review is focused on the connections between inflammation and MEC pathogenesis. These connections can be used as attack points for pharmaceuticals. For this we summarized the results of research undertaken over the last 30 + years. MEC pathogenesis can be described by specific inflammatory dysregulation already known from arthritis, Crohn's disease or multiple sclerosis. A hallmark of this dysregulation are positive feedback loops of the inflammation further amplifying itself in a vicious circle-like manner. We have identified over one hundred drugs which are already used in clinic to treat other inflammatory diseases, and could potentially

*Correspondence: holger.sudhoff@uni-bielefeld.de

¹ Department of Otolaryngology, Head and Neck Surgery, Universität Bielefeld, Teutoburger Str. 50, 33604 Bielefeld, Germany Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/jublicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

be repurposed to treat MEC. To improve and expedite clinical success rates, we applied certain criteria based on our literature searches and condensed these drugs down to the 13 top drugs. We hope the review will serve as ground-work for the primary goal, which is to provide potential pharmaceutical therapies to MEC patients for the first time in history.

Background

Middle ear cholesteatoma: A disease without medical treatment

The middle ear cholesteatoma (MEC) is an inflammatory, destructive, and locally invasive middle ear lesion composed of proliferative keratinizing squamous epithelium and its subepithelial connective tissue. In general, there are two kinds of MECs. The rare congenital form arises from squamous epithelium trapped within the temporal bone during embryogenesis, and the more frequent acquired type of MEC which has an annual incidence between 7 and 15 out of 100,000 [1-4]. As disease progresses, the lesion becomes chronically and increasingly inflamed and this escalating inflammation leads to several complications. (1) Erosion of adjacent bony structures resulting in conductive and/or sensorineural hearing loss. MEC may lead to severe complications like (2) vestibular dysfunction, (3) facial palsy, (4) brain abscesses and other intracranial complications. Even though this severe disease was first described by Duverney in 1683, there is still no medical treatment, hence surgical removal of the MEC offers the only therapeutic possibility today [5]. Unfortunately, recurrence often occurs within 10 years in 40% [1] with even higher rates in children [6]. Rates of recurrent MECs are partly dependent on surgical techniques [7] and impacted by different approaches and techniques [8].

The main characteristic of MEC tissue is its abundant and chronic inflammation. It has been shown that the bone erosion as well as progression and recurrence are associated with the high degree of inflammation, which is also frequently linked to infection of the MEC tissue [9].

Applicable targets in precision medicine for MEC therapy

Methodology

Since chronic inflammation is the major source of MEC pathogenesis, this review will focus on the inflammatory intra- and extracellular signaling responsible for the etiology and pathogenicity of MEC. For this we use peer reviewed original research articles written in English as well as published review articles from the PubMed/Med-line database between January 1990 and April 2022.

In the first section, we will focus on the signs triggering the primary inflammation as well as the intracellular signalling and the inflammatory output of the immune system residing in the MEC niche. In the second section we discuss the mechanisms by which inflammation drives osteolysis, epidermal proliferation and angiogenesis. All of this is aimed at the identification of molecular targets which can be applied in precision medicine. For this we only sum up studies which were able to draw connections between these targets and MEC pathogenesis. Wewill distinguish between in vitro, animal models and clinical data to judge the relevance of a specific cue/factor to the clinical outcome. In vitro evidence were derived from primary cultures mostly derived from MEC tissue, in vivo correlations were exclusively obtained from well-established animal models of MEC. Clinical data was derived from either cohort studies, used to detect the correlations between a certain target and the clinical outcome, or controlled clinical trials, always used to prove deregulation of this target in different control tissue compared to MEC. During this review we will focus solely on these fractions of the signaling pathways which are deregulated in MEC tissue and hence more accessible to pharmaceutical treatment compared to the surrounding healthy tissue. To facilitate the main goal of this review which is the identification of possible pharmaceutical treatment options, we consider only targets for which previous investigations showed a non-contradictory connection to MEC pathogenesis and those without available drugs within our inclusion criteria (Additional file 1: Table S1), described in the subsequent paragraph. Based on these descriptions, we will discuss possible medical treatments, aimed at the described roots and try to judge their potential to restore the proper function of the wrongly trained inflammatory network and their clinical potential. Indeed, based on the hierarchy of evidence described above, we derive a table marking the different drugs in accordance to the level of evidence. In this table we will suggest drugs, which have already been used for and applied to the middle ear and might be directly tested in clinical studies. Most importantly we proposed numerous drugs in the context of a molecular targeted therapy (MTT), which are already applied systemically in clinic but might previously be tested for their pharmaceutical effect on MEC in an animal model. For some targets numerous drugs are available. In this case we have suggested the ones which are already approved for clinical application. To enable fast translation into clinical use, the vast landscape of experimental modulators of inflammatory signalling only applied through an in vivo or in vitro model so far were ruled out. Also excluded were approaches based on RNA, lytic viruses or antibodies. Even though we think their topical application to an open wound-like tissue e.g. MECs might potentially work, this has never been tested and hence offers no fast track to translation.

By this we hope the comprehensiveness of this review inspires and streamlines new research into this topic, which might expedite the development of the first medical treatment option based on precision medicine, and effectively manages progression and recurrence of this hazardous lesion in numerous MEC patients.

Targets from initial triggers of inflammation

The chronically escalated inflammation in MEC is known as a predictor for its pathogenesis and pathogenicity. Hence, in vivo studies have showed a decrease of MEC pathogenesis by systemic application of prednisolone used to treat various inflammatory conditions [10, 11]. Due to its ototoxic properties, these kinds of drugs are only applied topically and systemically in complicated cases with labyrinth fistula, brain abscess or meningitis [12]. But more recently the non-ototoxic drug montelukast usually used for maintenance treatment of asthma or allergic rhinitis [13], showed similar benefits. Therefore, a clinical study comparing local application of montelukast to traditional treatment, may enable new treatment options.

As already mentioned, this review focuses on a more targeted approach aimed at the characteristics of MEC disease. The main aspects of MEC tissue are tissue destruction, cell death as well as microbial infection. From these three sources the initial drivers of inflammation in MEC are derived. They can be grouped into the pathogen associated molecular pattern (PAMPs), which originate from bacteria, and damage associated molecular pattern (DAMPs), derived from stressed and dyeing cells (endogenous) and decomposing extracellular matrix (ECM; exogenous). Different studies have linked the PAMP Lipopolysaccharide (LPS), which originate from Gram (-) bacteria, to MEC formation [14] in patients, in vitro or in vivo MEC growth [15, 16], respectively and bone resorption in the clinic [14] as well as in an animal model [16]. But even though MEC often occur after chronic infection of the middle ear, clinical sampling has showed, that from nearly 40% of MEC, no bacteria could be cultivated [17]. But more recent studies comparing metagenomics approaches with conventional culturebased techniques, have demonstrated an improvement of the bacteria detection rate to 100%, hence MEC cannot be considered sterile [18, 19]. We assume, that even though MEC tissue is not sterile, the exact composition of the microbiome may define and contribute to the clinical outcome. For example, the concentration of PAMPs like lipopolysaccharide (LPS) in MECs ranged over 4 orders of magnitude [14]. Even though fungal infection of the middle ear is a very rare entity, which is almost exclusively seen in immunocompromised patients, the microbial genus most often detected in these studies has been found to be Aspergillus, and this fungus was most frequently found by culture based approaches [20, 21]. Interestingly, Aspergillus presence could also be linked to severe bone erosion in patients [18]. In any case, other mechanisms triggering the initial inflammation (particularly in the non-infected MECs) must also be present. Since tissue damage is characteristic for MEC disease, various DAMPs are upregulated and could be linked to osteolysis in an animal model [22, 23].

The prevention of the generation of endogenous or exogenous DAMPs is directly coupled to prevention of MEC. Hence, a drug inhibiting MEC formation will additionally cause a negative feedback loop further reducing DAMP driven inflammation. PAMPs on the other hand, are directly linked to the presence of pathogens and are responsible for the signalling through the macrophage inducible Ca2+-dependent lectin receptor (Mincle), which is able to detect PAMPs derived from fungal or bacterial sources, or Toll-like receptor 4 (TLR4) responding to bacterial PAMPs both characteristic for MEC. For bacteria, empirical broad-spectrum antibiotics are commonly applied post-surgery to prevent postoperative infection [24-26]. Due to the rise in antibiotic resistant bacteria ("super-bugs"), recent studies have shown that the application of specific drugs according to antibiotic susceptibility testing improved the rate of dry ear after surgery significantly [27, 28], a predictive factor directly coupled to MEC recurrence and bone destruction [29, 30]. This suggests that a clinical study comparing broadspectrum antibiotic treatment vs antibiotic susceptibility testing based treatment with longer follow-up periods will be able to identify parameters like recurrence, bone destruction etc. and may encourage surgeons to utilize antibiotic susceptibility testing before surgery. Since fungal infection and Mincle are coupled to MEC pathogenesis as well, application of antimycotics might yet be an additional experimental arm in this study.

Unfortunately, not only certain DAMPs and PAMPs are upregulated but their receptors are also expressed at a higher level compared to healthy tissue, further exaggerating the inflammatory signalling in MEC tissue. A major contributor in terms of DAMPs is the upregulated receptor for advanced glycation end product (RAGE) [31], for which S100A8/A9 and particularly HMGB1 can serve as ligand, both known to be upregulated in MEC [31–33]. In vitro studies could therefore potentially prove, that the adverse combination of the receptor and agonist upregulation of the HMGB1/RAGE pathway, contributes to MEC pathogenesis [31].

The DAMPs and PAMPs recognizing TLR4 have been found to be of particular importance in regards to MEC disease. Hence various investigations utilizing transcriptomic and proteomic techniques were undertaken and demonstrated an upregulation of TLR4 in comparison to various different control samples e.g. "normal" control middle ear samples without any inflammation [34], congenital MEC [35] or external auditory canal skin [36–38] and even normal tissue from chronically infected middle ear [37]. As already mentioned TLR4 does not only recognize DAMPs already upregulated e.g. HSP27 [39], HSP60 and HSP70 (both 40), HMGB1 [31] and S100A8/ S100A9 [32, 33] but can also be activated by PAMPs derived from Aspergillus and Candida or LPS, which are particularly abundant in purulent MECs [14, 21]. Unfortunately, an investigation utilizing Förster resonance energy transfer (FRET) measurements could detect a further amplification of the TLR4/LPS signalling through HSP70 by trafficking TLR4/LPS to the Golgi apparatus [41]. Various in vitro assays have suggested that TLR4 signalling is involved in the establishment of the inflammatory environment of the MEC niche [42] and MEC recurrence and proliferation [43]. Clinical data, as well as a Knock-out mouse model of acquired MEC directly linked TLR4 to the local inflammation and bone destruction [35]. TLR4 signalling is further enhanced by the triggering receptor expressed on myeloid cells-1 (TREM-1), which is upregulated compared to healthy control skin [44]. TREM-1 senses DAMPs like HMGB-1 and further amplifies the already exaggerated TLR4 signalling in vitro [45]. A second way to enhance the TLR4 signalling in MEC, might be the presence of macrophage migration inhibitory factor (MIF). Since MIF is able to induce TLR4 expression [46], the recurrent and osteolytic features of MEC disease may also be linked to the expression of MIF [47]. Due to the parallel activation of TLR4 and RAGE signalling in MEC and the interconnection between RAGE and TLR4 signalling on the intracellular level, these signalling inputs are able to bilaterally enhance each other. In this context, antibody blocking assays and animal models have shown that HMGB1 induced RAGE signalling increases the inflammatory potential of LPS in macrophages, one of the main immunomodulatory cells in MEC tissue [48]. A study which investigated various other PAMP recognizing receptors at the mRNA level found that Mincle, which is able to recognize PAMPs of bacterial or fungal origin respectively, is upregulated in comparison to external auditory canal skin and that its expression could be correlated to recurrence and osteolysis in MEC disease [49]. There are many more PRR receptors which are expressed in MEC tissue. But in contrast to TLR4, TREM-1, RAGE and Mincle, their contribution to MEC pathogenesis has not been proven so far.

Since it is hard to prevent PAMPs and particularly DAMPs in MEC, an inhibition of activation of PRR should be one of the first targeted approaches (Fig. 1). As described above TLR4 signalling is paramount in MEC pathogenesis. Interestingly, amitriptyline, a drug usually administered against migraine, inhibits the expression of TLR4- receptor and subsequent NF-kB signalling [50]. The up regulation of TLR4 expression, through MIF, might be dampened by Ibudilast known to inhibit MIF [51] and already applied against progressive multiple sclerosis [52]. Also, the MIF receptor antagonists RTL1000 [53] was evaluated for treatment of MS [54, 55]. After downregulating TLR4 expression the prevention of the activated (TLR4-MD-2-PAMP/DAMP)₂ complex is the next pharmaceutical target. In vitro studies from our group, showed that LPS based MD-2-TLR4 antagonists might be one pharmaceutical approach against MEC [42, 43]. The Antagonists eritoran can be used to treat septic shock [56, 57], JKB-122 against hepatitis [58], VB-201 for the therapy of psoriasis [59], ibudilast [60] or dalcetrapib used for cholesterol management [61] and curcumin, for which topical application was shown to be effective against some inflammatory skin disorders [62]. An MD-2-TLR4 antagonist established in clinical practice are Taxanes [63] known from oncologic therapies [64]. But even though the (TLR4-MD-2-PAMP/DAMP)₂ might be formed, targeted inhibition of downstream (MyD88dependent and MyD88-independent) pathways offers an additional anti-inflammatory strategy. Resatorvid is a selective inhibitor of TLR4 which inhibits autophagy and is used for various inflammatory situations, which include neuroprotection following brain injury, the treatment of severe sepsis [65] or alcohol-related cirrhosis, and also inhibits both MyD88-dependent and MyD88independent pathways [66]. The compound resveratrol used against arthritis or COPD [67] and alogliptin against Typ-2-Diabetes [68] blocked MyD88-dependent signalling [69, 70], respectively. Naltrexone and naloxone, drugs well known as opioid antagonists but also applied against inflammatory disorders like Crohn's disease and neuroinflammation respectively, are MyD88 independent inhibitors of TLR4 signalling [71]. To inhibit the amplification of TLR4 signalling through TREM-1, its transcription can be inhibited by curcumin [72], or the blocking of TREM-1 ligand binding domain can be achieved by peptides like nangibotide already applied against septic shock [73–75]. RAGE can be targeted by extracellular inhibitors e.g. azeliragon applied against Alzheimer disease [76]. To target Mincle, quercetin, effective against upper respiratory tract infections [77], or curcumin [78], were shown



to inhibit Mincle induced macrophage inflammation [79].

Targets from extra- and intracellular inflammatory signalling pathways

As a consequence of the upregulated PRR-signalling, a proinflammatory environment is created with IL-1 α , IL-6 and TNF- α being highly upregulated in the inflamed MEC tissue [80, 81]. Apart from the elevated expression of IL-1 β , the corresponding Interleukin-1 receptor (IL-1-R) is upregulated threefold in MEC epidermis compared to the normal aural skin [82]. This pro-inflammatory signalling induces several intracellular signalling events characteristic for MEC tissue. The main transcription factors involved in this are NF- κ B, AP-1 and HIF-1. Investigations have showed, that the expression of NF- κ B was upregulated and more importantly the

nuclear translocation, as well as the DNA binding activity were increased in epithelium of MEC tissue compared to healthy skin [83]. Furthermore, a significant correlation was found between TLR4 and activated NF-κB [37] in MEC tissue, and we assume, that the same could be shown for RAGE as well. Further, another study demonstrated a correlation between NF-kB and biomarkers of angiogenesis and osteolysis in MEC tissue isolated from patients [84]. In addition to this, the DNA-binding protein inhibitor Id1, known to be induced by chronic TLR4 signalling [85], was proven to be abundantly expressed in MEC epithelium [86] where it accelerates proliferation and survival of keratinocytes in vitro via NF-KB signalling [87]. Two studies found that the transcription factor AP-1 was detectable in the nucleus of MEC epithelium [88], where it was much more abundant compared to healthy ear canal skin [89].

Since hypoxia is common in MEC tissue, the expression of hypoxia-inducible factor 1 (HIF-1) was significantly increased in MEC compared to normal skin [90] and localized to keratinocytes located on the basal layer [91]. Interestingly, the likelihood of relapse after surgery was correlated to HIF-1 expression [90]. Further, downstream of the PRR and cytokine receptors, the transcription factor NF-KB is the most promising contributor to angiogenesis and osteoclast activation in MEC. Since NF-κB is a common target of various diseases, numerous inhibitors have been designed and clinically tested (reviewed in 92). Even though the role of AP-1 has not been not elucidated so far, AP-1 shares many upstream inducers with NF-KB. Hence both transcriptional factors can be inhibited by the same signalling transducing protein inhibitors (Fig. 1). Particularly interesting drugs for MEC are inhibitors of Bruton's Tyrosine Kinase (BTK), since BTK lies downstream of TLR4, TREM-1, IL-1-R and TNFR, the dominant links between extracellular and intracellular signalling in MEC. The BTK inhibitors acalabrutinib [93], ibrutinib [94], zanubrutinib [95] and tirabrutinib [96] have been developed and applied to treat oncologic diseases but also various inflammatory skin diseases [97].

Another protein downstream of TLR4 and IL-1-R is Interleukin-1 receptor-associated kinase (IRAK). The inhibitor PF-06650833 developed against rheumatoid arthritis [98] is useful for inhibition of IRAK. Downstream of TLR4 and TNFR is the cellular inhibitor of apoptosis proteins (c-IAP), which can be targeted by inhibitors like birinapant usually used against solid tumors [99]. Pi3K Phosphoinositide 3-kinases PI3Ks, are a family of enzymes downstream of TLR4, TNFR and IL-6-R. Their inhibition by idelalisib [100], copanlisib [101] or duvelisib [102] approved for the treatment of lymphoma and alpelisib [103] useful against breast cancer. Directly upstream of the activation of NF-kB is IKK a/b, of which the inhibitors CHS-828 [104] have been used against solid tumours and VGX-1027 [105] for the treatment of rheumatoid arthritis. In the canonical pathway, proteasome degradation of IkB is crucial for the release of active NF-kB. Many useful proteasome inhibitors have been already applied in clinical practice e.g. disulfiram against alcohol dependence [106], ixazomib [107], carfilzomib [108] and oprozomib to treat multiple myeloma [109] or marizomib developed against glioblastoma [110].

HIF-1 can be inhibited by several drugs already in clinical practice, which act on different levels (reviewed in 111, 112). The accumulation of HIF-1 is effectively targeted by either enhancing its degradation with the drugs panobinostat [113], vorinostat [114], geldanamycin [115] and its derivative tanespimycin [116] applied against solid tumours and haematological malignancies or resveratrol used to treat against arthritis or COPD [117]. Another approach would be the inhibition of its expression with EZN-2208 [118] or EZN-2968 developed against solid tumours and lymphomas [119] or panzem which was additionally used in rheumatoid arthritis [120]. Interestingly panzem reduces HIF-1 transcriptional activity as well, hence it could attack HIF-1 signalling on two levels. Despite promising results, clinical trials with panzem were dismissed, due to its poor oral bioavailability. But topical application to MEC might revive the clinical prospects for this promising drug. Alternatively, inhibition of HIF-1 transcriptional activity can be achieved by echinomycin [121] or amphotericin B [122], known for its antifungal effect additionally reducing Mincle signalling. Finally, the drug acriflavine [123] can be used to dampen HIF-1 signalling by preventing HIF-1a/HIF-1b dimerization and is also able to inhibit the growth of Gram (+)bacteria as well, reducing microbial driven inflammation on two levels.

Even though Id1 can only be linked by in vitro models to epidermal proliferation as well as angiogenesis of MEC, it might be inhibited by many clinical approved drugs (reviewed in 124). Chief among these were fucoidan [125] and berberine [126], both inhibiting the expression of id1 and showing good effect against inflammatory disorders.

In conclusion, it is clear that numerous PAMPs/ DAMPs and cytokines are abundant in MEC tissue and can be linked to promotion of MEC characteristics via establishment of an inflammatory environment. Importantly, only signalling via LPS/TLR4/NF- κ B was demonstrated to contribute to the initial establishment of the inflammatory state of MEC disease as well as to its clinical outcome. In accordance with our criteria judging the quality of a target, we suggest that this pathway might be the most promising pharmaceutical target.

Specific targeting of Innate immunoregulatory events in MEC disease

Due to its highly inflamed sub-epithelial connective tissue, middle ear MEC is characterized by a large infiltration of inflammatory cells. Macrophages are the subset of myeloid-derived blood cells defined here precisely as CD3+ and CD68+ cells, which have been found to be upregulated by a factor of about five in MEC showing high level of bone destruction compared to external auditory canal skin [127]. Other studies utilizing MEC tissue of all levels of invasiveness, have detected a two to fourfold increase of CD11c and CD68 positive cells in relation to normal ear skin and characterized them as dermal macrophages [127, 128]. We hypothesize that the polarization of MEC macrophages is likely to be of the proinflammatory M1 type, which is demonstrated by the high expression of nitric oxide synthase [129] and proinflammatory cytokines [80] in MEC. Another myeloid cell type, dendritic cells, were also increased in acquired MEC compared to congenital MEC [130] and their presence could be linked to drivers of osteolysis in MEC tissue in an animal model [131].

After infiltration, the differentiation of monocytes into proinflammatory M1 macrophages and conventional dendritic cells should normally be inhibited. There are several drugs already applied in clinic able to inhibit this polarization of macrophages in inflammatory disorders e.g. curcumin [132], bilobalide used against complications in pregnancy [133], quercetin already applied against COPD [134, 135], berberine [136] or rosiglitazone [137] known for their positive effect on diabetes, arctigenin [138] which is suspected to lower the risk of exaggerated inflammation. In regards to the dendritic cells abundant in MEC, substances like corticosteroids, long known for their application against inflammatory disorders [139], rapamycin, an immune modulator useful in helping prevent transplant rejection [140] or cyclosporine [141], known as a potent T-cell inhibitor applied in transplant situations where high immunosuppression is required, but also used against psoriasis, are capable of inducing the differentiation into tolerogenic dendritic cells priming the immune system into tolerogenic state against various antigens. Another myeloid cell type, the mast cells, are known to influence multiple features of persistent inflammation (reviewed in 142). In accordance with that, investigations showed a three-to sevenfold increase in MEC in relation to normal epidermal tissues [143], which are localized close to the epithelium [144] and to the eroded surfaces of the bone [145]. Since mast cells are known to influence various aspects of bone absorption, and particularly osteoclastogenesis, by histamine (reviewed in 146), we suggest they also increase osteolysis around MEC. In addition the released histamine is known to favour a proinflammatory response [147] further enhancing this phenotype in MEC tissue.

Due to its impact on the symptoms of inflammatory diseases, various drugs have been developed to inhibit the degranulation of mast cells e.g. in allergic conjunctivitis (reviewed in 148). The most popular are sodium cromoglycate, also applied against asthma and allergic rhinitis or the more potent lodoxamide, nedocromil as a prophylactic in asthma, ketotifen is also used against inflammatory conditions like allergic rhinitis and conjunctivitis, and chronic spontaneous urticarial. and the popular olopatadine additionally useful to treat allergic rhinitis. The subsequent inflammatory response can be inhibited by the vast numbers of clinically approved H1-antihistamines [149, 150] acting mainly on Th1 cells and M1 macrophages [151] with particularly the second generation having less side effect on the nervous system and should be preferred in middle ear application. Even though usually applied to treat hay fever, cetirizine, fexofenadine and loratadine were also effective against urticarial diseases and might be a treatment opportunity for MEC as well. Interestingly, ketotifen and olopatadine work as histamine 1 receptor antagonists as well and hence might attack histamine on two levels [148]. The clinically approved H2-antihistamines like nizatidine are known to reduce exaggerated inflammation after burn injury [152], and mainly reduce Th1 driven inflammation while newer H4-antihistamines (reviewed in 153, 154) mainly act on the mast cells to prevent their activation. Hence H4-antihistamines like toreforant have been used in asthma and psoriasis [155], JNJ39758979 used in clinical trials to treat symptoms of atopic dermatitis [156], ZPL-3893787 tested against atopic dermatitis and psoriasis [157] and UR-63325 developed to treat atopic dermatitis [156] might offer new opportunities.

Pharmaceutical targets within the lymphoid cell regulation in MEC

As regards lymphoid lineage cells in the MEC, T lymphocytes are much more abundant in acquired MEC tissue compared to congenital MEC [158] or normal ear skin [127, 159]. Different studies aimed at delineating the Th1/ Th2 paradigm of inflammatory events in MECs, found that according to the expression profile showing TNF- α , IL-1 β and IL-6 and lacking IL4, IL4R, IL10 and IL10R, the Th₁ cells were likely to be dominant [80, 144].

The differentiation of T₀ cells into proinflammatory Th₁ cells and their subsequent activation is another major proinflammatory cellular process in MEC. The differentiation can be blocked by clinically approved drugs like beta2-agonists, a group of drugs with a long history prescribed for asthma [160, 161] (with a wide range of different drugs) (reviewed in 162). Other drugs include progesterone applied against endometriosis [163], glucocorticoids known for their potency in suppressing various allergic, inflammatory, and autoimmune disorders [164], flavocoxid which has proven useful against osteoarthritis [165], epigallocatechin gallate used to treat multiple sclerosis [166] or phosphodiesterase-4 inhibitors regularly utilized in the case of inflammatory diseases like asthma, COPD, atopic dermatitis and psoriasis [167–169]. To reduce the activation of Th₁ cells and inhibit Th₁ cytokine secretion, the clinically approved drug Vitamin D3 [170] known as treatment primarily against osteoporosis (or any cause of Vit D deficiency) can be used.

The large amount of infiltrating cells in MEC tissue is likely due to the overexpression of the intracellular adhesion molecule-1 (ICAM-1) found in vessels of acquired MEC compared to normal skin [171], congenital MEC [158] and ear canal skin, tympanic membrane or facial skin (all [172]. Studies have linked the high expression of ICAM-1 to the level of inflammation in MEC [158] and in vitro studies have demonstrated that human dermal microvascular endothelial cells expressed ICAM upon stimulation with typical M_1/Th_1 cytokines e. g. IL-1 α , IL-1 β , and TNF- α [173] all of which are highly upregulated [81, 174]. We assume that this establishes a spiral of doom of self-enhancing inflammation, in which there is enhanced migration of Th₀ and M_0 cells into the tissue, their differentiation into M_1/Th_1 cells further increases the inflammation, and this upregulates ICAM-1 which again results in even more immune cell migration and inflammation respectively.

The enhanced expression of ICAM in MEC endothelium cells, responsible for increased infiltration of MEC tissue, might be inhibited by several clinically applied drugs like resveratrol [175] or another rheumatoid arthritis drug called methotrexate [176], curcumin [177] and especially alicaforsen which showed promising results against pouchitis and different kinds of inflamed colorectal conditions [178].

Since T-cells seem to play a key role in the progression of inflammation in MEC disease, the role of Human Leukocyte Antigen-DR isotype (HLA-DR) was also more closely investigated in the context of MEC disease. In general, HLA-DR functions as MHC class II cell surface receptor presenting e.g. bacterial antigens to T-cells. HLA-DR is upregulated [144, 159] and abundantly expressed in macrophages present in the perimatrix of MEC tissue [179, 180]. The high expression of HLA-DR on macrophages in combination with the abundance of e.g. bacterial antigens results in the activation of the frequently found T-cells and is thought to be an additional source of the immunologically activated state of the infiltrating cells observed in MEC tissue [180, 181]. To reduce HLA-DR expression, Iscador usually applied against solid tumors can used [182].

It can be said definitively that the M_1 and Th_1 cells residing in MEC tissue are major contributors of inflammatory cytokines and antigen presentation via macrophages and dendritic cells to Th_1 cells is the most frequent immune regulatory event in MEC tissue (Fig. 2). All this is crucial in maintaining the chronic inflammatory state of the MEC niche responsible for the clinical symptoms described in the upcoming chapters.

All the approaches mentioned above target the reduction of the proinflammatory MEC signalling initially triggered by PRR and amplified by proinflammatory extracellular signalling. In the second section we want to focus on the two major symptoms of MEC disease the osteolysis and the uncontrolled proliferation of keratinocytes and MEC mass supported by angiogenesis and their applicability as target in MEC therapy. We want to remind our readers, that this strategy not only targets the symptoms but more importantly, eliminates the "fuel to the fire" of the ongoing inflammation, the DAMPs derived from keratin debris and bone matrix.

Directly targeting links between inflammation and drivers of osteolysis

As mentioned above conductive hearing loss is a frequent and serious complication caused by MEC disease. It can result from tympanic membrane rupture and/or changes in the ossicular chain due to fixation or most often bony erosion caused by the chronic inflammatory process [183]. Today there are two different mechanisms suspected to be relevant in the process of bone resorption by MEC, the (1) abundant osteoclast activation and (2) enzymatic digestion. These mechanisms are triggered and enhanced by the severe inflammation in a positive feedback manner as discussed. In the upcoming section we will highlight the main influences behind these mechanisms and their contributions to the clinical outcome.

A previous histo-pathological study detected an increase in osteoclast density [81] and other studies have directly linked osteoclast activation to bone destruction found in MEC disease [81, 184]. For osteoclast activation, the receptor activator for nuclear factor kappa B ligand (RANKL) has been identified as a key factor in promoting the differentiation and fusion of osteoclast precursor cells and activating the bone resorption by mature osteoclasts, while osteoprotegerin OPG on the other side is a decoy receptor for RANKL negatively regulating this process [185]. In accordance with this RANKL-positive cells and/ or the ratio of RANKL/OPG were significantly higher in MEC compared to every other skin sample e.g. external meatal skin [186, 187], post auricular skin [188] normal skin [189], auditory canal skin and even granulation tissues [190] and RANKL expression positively correlated with osteolysis in MEC patients [186, 191]. In contrast to the upregulation of RANKL in the soft body tissue adjacent to the decomposing bone, there was no evidence found for an increase in RANKL or RANKL/OPG ration inside the bone tissue more distant to MEC [192]. Various investigations have tried to shed light on the source of RANKL in MEC. They found, that overexpression of RANKL is typically expressed by activated T-cells [193, 194] abundant in the perimatrix, but also from epithelial [186, 191] and fibroblastic cells [81, 186].

A paracrine positive regulator of RANKL signalling in MEC, the parathyroid-hormone-related protein (PTHrP), is usually expressed in keratinocytes of the skin. Since PTHrP is known as an inflammatory marker [195], keratinocytes from MEC express more



PTHrP than normal keratinocytes [196]. Most recently a study demonstrated, that not only protein expression of PTHrP in MEC epithelium is significantly increased but also positively correlated with the expression of RANKL in osteoblasts [197] and correlated to the degree of bone resorption [191]. Another source of RANKL is known from rheumatoid arthritis, where the RANKL-Th17 system contributes to bone destruction in a paracrine manner. Another study showed that IL-17 immunoreactivity was increased and was localized to CD4+ lymphocytes. Most importantly this correlated with the number of cells positive for RANKL and the degree of bone destruction [198]. We assume that TNF- α and INF- γ secreted by the numerous Th₁ cells triggers the production of IL-23 in macrophages, and IL-23 ultimately differentiates Th₀ into Th₁₇ cells which further enhances osteolysis in MEC disease, by priming fibroblasts and maybe osteoblasts with IL-17 and thereby inducing RANKL in a paracrine manner.

As described in the previous paragraphs, the main driver of osteolysis is RANKL derived from inflamed keratinocytes, fibroblasts and Th₁₇ cells. Inhibition of RANKL signalling in clinical trials by small molecules was achieved by bisphosphonates developed against osteoporosis [199], AZD4547 a new anti-cancer drug [200], isoflavone (useful against asthma) [201], and iguratimod, used to treat rheumatoid arthritis [202]. The widely used bisphosphonates are able to suppress RANKL expression [203]. PTHrP derived from epithelium inducing RANKL expression in osteoblasts might offer an alternative target in the MEC context. The clinically approved antitumor agent cabozantinib decreased PTHrP expression and might be useful in this context [204]. The secretion of another factor responsible for osteoclast differentiation, IL-17, can be reduced by clinically applied drugs KD025 [205, 206], fused pyrimidines [207] or iguratimod [208].

In vitro studies on primary MEC epithelia cells demonstrated, that expression of RANKL can be induced by stimulation of the TLR4 by LPS [209]. However, since clinical data demonstrated no correlation between RANKL expression and bacterial infection of the MEC [17], the correlation between osteolysis and LPS, as well as TLR4 might have another source. Hence the connection between proinflammatory cytokines and osteolysis in MEC was further investigated in various studies. In vitro studies showed, that LPS initiated expression of TNF- α , IL-1 β and IL-6 via TLR4 pushed the precursor cell in an autocrine manner, and primed with sub-osteoclastogenic doses of RANKL, into fully differentiating phases [210]. This is in accordance with the clinical finding, that IL-1 α , IL-6 and TNF- α are highly upregulated in the inflamed MEC tissue and known to contribute to osteoclast activation (reviewed in 211). As a result, studies could correlate TNF-a, IL-1a as well as IL-6 directly to the severity of bone destruction in MEC patients [80, 212-214]. In accordance with this, an in vivo model of MEC linked the presence of proinflammatory macrophages and T-cells to bone resorption of MEC [22, 23]. This cytokine driven maturation might explain the observed correlation between bone resorption and LPS [14], without induction of RANKL by LPS [17]. The central role of TLR4 was further confirmed by an TLR4 knockdown in an animal model of acquired MEC, which reduced the amount of osteoclast formation and bone destruction significantly [35].

Other cytokines were investigated as well but no clear evidence was found for their correlation to bone resorption in the clinic. For example the macrophage-colony stimulating factor, crucial for complete osteoclastgenesis (reviewed in 215), was found to be overexpressed in MEC specimens compared to normal external meatal skin [193] in Th₁ cells [193] and IL-1 β known to promote osteoclast activation [216] and LPS induced bone resorption in animal models [217], is also upregulated [81]. But no further investigations were undertaken to elucidate the role of M-CSF or IL-1 β driven osteolysis in MEC.

Different drugs are available to target these cytokines. The processing of IL-1 β can be inhibited by several inflammasome inhibitors (reviewed in 218). Some of them have already been applied clinically i.e. parthenolide to treat various inflammatory diseases [219], pralnacasan [220] or dapansutrile [221] to treat rheumatoid arthritis and tranilast shown to be useful against asthma and keloid scars [222]. For the IL-1 receptor downstream signalling, diacerein used against osteoarthritis in clinic might be a useful approach [223]. Clinical inhibition of IL-6 signalling is achieved by several clinically approved JAK inhibitors from the 1st and especially the more specific 2nd generation, as well as the STAT3 inhibitors (reviewed in 224). The most popular among these drugs are upadacitinib used in patients with inflammatory driven diseases across gastroenterology, dermatology and rheumatology [225], filgotinib [226] or peficitinib [227] efficacious in patients with rheumatoid arthritis and the STAT3 antisense oligonucleotide AZD9150 [228] applied against cancer. As regards TNF- α , numerous small molecules are suspected to be effective [229] but only Iguratimod [202] was developed to the clinical stage testing where it was used to treat rheumatoid arthritis [230]. M-CSF is suspected to play an important role in osteoclastogenesis in MEC, and M-CSF can be decreased by bisphosphonates [199] and the pharmaceutical AZD4547 [200]. M-CSF signalling might be further decreased by the novel M-CSFR inhibitor pexidartinib which has been clinically approved for cancer therapy [231].

Even though it is known that COX-2/PGE2 plays a central role in osteoclast activation in general and in LPS induced osteoclast activation in particular [232] it has not been investigated in MEC driven osteolysis so far. Upregulated expression of the enzyme COX-2 needed for PGE2 synthesis has been localized in MEC epithelium [233]; the proinflammatory prostanoid receptors EP1-EP3 are upregulated while the anti-inflammatory EP4 receptor is downregulated [234] which suggests an enhanced proinflammatory PGE2 signalling in MECs. Numerous well characterized COX-2 inhibitors are available and in particular, the ones specific to COX-2 might offer a treatment opportunity for MEC. Among them are drugs like celecoxib [235], etoricoxib [236], rofecoxib [237] and meloxicam [238] all proven to be useful against different forms of arthritis.

The presence of matrix metalloproteinases (MMPs) might be a driving force behind the enzymatic digestion of bone in MEC disease. The presence of MMPs play an important role in resorption of the bone by decomposing the organic components of bone tissue. In accordance with the well documented induction of MMP-2 and MMP-9 via NF- κ B and AP-1, the expression of MMP-2 as well as MMP-9 is generally upregulated in comparison to healthy skin and correlated to the level of inflammation [239–242]. The high expression of MMP-2 was almost exclusively found in fibroblasts or in the inflammatory cells of the perimatrix [243] and positively correlated to bone destruction in patients [212, 239] and the expression of MMP-9 was found to correlate to the level of osteolysis as well [212, 244].

Due to their attractiveness as pharmaceutical targets, numerous inhibitors of MMPs were developed in the past and tested in clinical trials [244–246]. All these drugs were applied on solid tumors to prevent angiogenesis or metastasis e.g. chlorotoxin inhibits MMP-2 [247] or broad spectrum inhibitors acting on MMP-2 and MMP-9 simultaneously like diazepinomicin [248], rebimastat [249], marimastat [250], arctigenin [251]. Unfortunately, many drugs have failed safety trials predominantly due



to adverse effects coupled to their systemic use, hence it is logical to assume a topical application may resurrect numerous drugs that have failed to reach the market. Interestingly, in vitro experiments with Vitamin D3 showed a decrease of MMP-2 and MMP-9 in inflamed fibroblasts [252] and in MEC Keratinocytes [253]. Since Vitamin D3 might has been used therapeutically against different forms of arthritis [254], it might offer an additional and much safer therapeutic opportunity.

It can be said definitively, that a spiral of doom leading to osteolysis is established in MEC tissue. In highly inflamed MEC tissue the expression of various osteoclastic activators TNF- α /IL-1 α /IL-1 β /IL-6/RANKL/ PTHrP/M-CSF are induced initially through TREM-1 fuelled TLR4 signalling and subsequently in a paracrine manner. This leads to activation of osteoclasts and subsequent osteolysis. This bone lysis is further amplified by the expression of MMPs in the inflamed MEC niche. As observed for other inflammatory osteolytic diseases (255, reviewed in 256), the DAMPs released from the bone tissue intensifying the TREM-1/TLR4 signalling even more and further boosting the osteoclast activation and MMP secretion (Fig. 3).

Targeting the fractions of the inflammatory network inducing epidermal proliferation

The most prominent feature of MEC disease is the ongoing proliferation of keratinocytes and the accumulation of keratin debris. Several factors are upregulated in MEC tissue due to its inflammatory state and are known to stimulate the proliferation of MEC keratinocytes e.g. PGE2 [15, 257], TNF- α [258], IL-1 α [259] or IL-6 [260]. To reduce the effect of these proliferative factors, PGE2, TNF- α , IL-1 α and IL-6 can be targeted by the approaches described in the previous subsection.

The most important stimulation of epidermal proliferation in MEC tissue is induced through paracrine signalling, via a mechanism well known from cutaneous wound healing and skin tissue homeostasis. In this process keratinocytes express PTHrP as well as IL-1 α and IL-1 β . These proteins will induce the expression of epidermal growth factor (EGF) and keratinocyte growth factor (KGF) in the fibroblasts of the dermis. These factors are vitally important in final induction of epidermal proliferation. Due to the inflammatory processes described above the expression of PTHrP [191], IL-1a [158, 174, 261],] and IL-1 β [81, 261] are already upregulated in MEC keratinocytes. The expression of IL-1 β might be further augmented by DAMPs activating the inflammasome as demonstrated for MEC keratinocytes [262]. In addition to the upregulation of the IL-1s, the interleukin-1 receptor antagonist (IL-1-RA) showed a decrease compared to healthy skin samples [263], further enhancing the effect. The expression levels of KGF [264, 265] as well as EGF [174] were also enhanced in fibroblasts residing in MEC tissue compared to fibroblasts of healthy skin. The high expression of EGF could be attributed to the fact that fibroblasts derived from healthy skin showed a much weaker expression of epiregulin [266] after in vitro stimulation with IL-1 α and/or IL-1 β compared to MEC fibroblasts. The increased expression of KGF in patients could be correlated with strong levels of the inflammatory infiltrate [267] and inflammation in general [268]. The paracrine signalling between fibroblasts and keratinocytes is further enforced by the increased expression of the corresponding receptors KGFR [241] and EGFR [269, 270] in the epithelium of MEC. In particular, KGF was found to be one of the main players in MEC development and able to induce MEC formation solely by its overexpression [271–274]. In accordance with that, the expression of KGF and KGFR could be directly correlated to epidermal proliferation of MEC tissue [241, 275, 276] as well as to proliferation of stem/progenitor cells [273, 274] which might also explain the correlation between KGF/KGFR and pathogenesis, as well as clinical recurrence of MEC after surgery [264, 275].

This suggests, that due to keratinocyte hyperproliferation and the high level of inflammation in MEC tissue, the fibroblasts receive excess PTHrP as well as IL-1 α and IL-1 β . This establishes a spiral of doom with increased paracrine signalling via EGF and KGF which further enhances the amount of PTHrP as well as IL-1 α and IL-1 β secreting keratinocytes and keratin debris which further increases inflammation by PRR signalling. All this makes this paracrine positive feedback loop spin completely out of control. PTHrP as well as IL-1 α and IL-1 β can be targeted by the means described in the previous subchapter. For KGF, the main driver of epidermal proliferation, targeted therapies via entrectinib [277] and larotrectinib [278] are available. The upregulation of different cytokines not only directly stimulates epidermal proliferation but more importantly fuels the paracrine feedback loop, which supplies growth factors to the keratinocyte. This will lead to more keratin debris and DAMP respectively, resulting in a self-enhancing inflammation. The significance of the contribution of hyperproliferation to MEC pathogenesis is reflected in the correlation between upregulation of Ki67 and downregulation of p27 relative to meatal skin tissues and the worsening of the prognosis in terms of bone erosion and recurrence rates [279]. In addition to this, paracrine signalling experiments on a complex in vitro model, which utilized a co-culture of fibroblast and keratinocytes derived from MEC, demonstrated an impressive reduction of viability and increase of apoptosis in karatinocytes after application of the COX-2 inhibitor [280].

Pharmaceutical targets responsible for angiogenesis in MEC

To support the described hyperproliferation, the abnormal growth of new vessels is crucial and accordingly the perimatrix of MEC contained significantly more micro vessels in the subepithelial connective tissue compared to ear canal skin [281, 282]. Since EGFR is upregulated [269, 270] and promotes angiogenesis [283], it is not surprising, that the increased density of micro vessels correlated with the increased expression of EGFR in inflamed MEC [269]. Due to its clinical significance, several small molecule inhibitors of EGFR signalling were developed and approved for clinical application. The most common are afatinib [284], neratinib [285], dacomitinib [286] and gefitinib [287] all approved to be effective against EGFR driven carcinomas.

Other drivers of angiogenesis have been investigated as well. The most prominent is the vascular endothelial growth factor (VEGF), for which expression is upregulated in compared to normal skin [241, 282]. A correlation between NF- κ B driven inflammation and VEGF was shown to promote the neovascularisation in MEC tissue [84]. VEGF signalling can be inhibited by approved drugs on the level of intracellular signalling by drugs like sunitinib [288] or axitinib [289], which are approved for treatment of renal cell carcinoma or nintedanib applied against pulmonary fibrosis [290] or lucitanib used in clinical trials against cancer [291].

Indeed other growth factors induced by inflammation and related to angiogenesis like bFGF [292] or TGF- β [293, 294] are also upregulated in MEC epithelium [295, 296], respectively. Since it became clear that stromal expansion plays a rather secondary role in MEC pathogenesis, these factors were neglected in MEC research. In the light of the paracrine positive feedback loop between epidermal and stroma cells, the stromal proliferation and these factors respectively may again be invoked as pharmaceutical targets.

Due to the role of FGF/FGFR in various diseases, numerous pharmaceuticals targeting the signalling pathways have been tested [297]. Even though only ponatinib is approved for the treatment of chronic-phase chronic



myeloid leukemia [298], there are several under clinical investigation in phase III trials, i.e. orantinib [299], cediranib [300] or lucitanib [291] used against cancerous disease or nintedanib to treat pulmonary fibrosis [290], interestingly the last two drugs work also as VEGFR inhibitors. For TGF- β inhibition only two clinical drugs are available so far, one is the antisense oligonucleotides AP-12009 which act directly against the mRNA of TGF- β [301] and the TGFB receptor antagonists like galunisertib, which are both used against carcinomas [302].

Other angiogenic factors e.g. IL-8 and COX-2 are highly expressed in the inflamed MEC tissue [38, 233, 303] respectively. An interesting study showed, that id1 increased endothelial proliferation in MEC by upregulating the expression of COX-2, VEGF via ERK1/2 and IL-8 through NF- κ B [304]. Hence drugs targeting id1 not only reduce epidermal proliferation but angiogenesis as well.

Other upregulated angiogenic factors upon inflammation are the already mentioned proteinases MMP-2, MMP-9. These MMPs are well known to be crucial for the decomposition of the extracellular matrix making the tissue susceptible for vascular outgrowth. An immunofluorescence correlation study in cholesteatoma tissue demonstrated, that correlations between NF- κ B and these MMPs causes the intensification of angiogenesis in MEC [84] and can be targeted by drugs described in the previous subchapter.

In sum, the increase of blood supply to the MEC is supported by its own inflammation by the upregulation of EGFR, VEGF, MMPs and presumably COX2, bFGF or TGF- β . This suggests, that this increased vascularisation by inflammation is a self-fuelling spiral of doom, since it not only enables the hyperproliferation but also provides better access for the T-cells, mast cells or monocytes/macrophages to the MEC further enhancing the inflammation (Fig. 4).

Conclusion

To date, numerous studies to inhibit the formation or recurrence of MEC have been undertaken. Unfortunately, many of these have focused on reducing the (already triggered) hyperproliferative behaviour of MEC epithelial cells, via cytotoxic approaches usually applied to cancer e. g. photodynamic therapy [305] or cytostatic



5-fluorouracil [306]. Even though 5-fluorouracil made its way into clinical trials [307, 308] the results were insufficient to establish its application in routine clinical practice for MEC. We suggest long term application of non-cytotoxic substances targeted on the inflammatory pathways described in this review might well be a superior strategy. In accordance with long term application, gradually absorbed formulations comprising non ototoxic carrier substances like hydrogels based on chitosan glycerophosphate and hyaluronic acid [309], are mandatory, but are not within the remit of this review. As regards topical application, the MEC site shows characteristics of an open wound rather than an intact epithelial barrier. Hence, topical application might work for many drugs normally not able to cross the intact epithelial barrier. Indeed, local topical application, would revive the prospect of many drugs described in this chapter, which showed severe side effects in clinical trials due to systemic application.

In this review we have proposed numerous molecular targets which appear to be vital to the deregulated inflammation in MEC, and known to be correlated to MEC pathogenesis. Unfortunately, the connection between certain targets e.g. AP-1 were not investigated in such detail, hence specific molecular targeted therapy (MTT) approach for them were not included. Other targets like TREM-2, GMCSF, etc. (Additional file 1: Table S1) were clearly correlated to pathogenesis, but for these no drug within the restrictions applied in this review were available up to now. Also the role of cytokines important for Th1 differentiation like IL-12, IL-18 or INF- γ with the last one being also crucial for M1 polarization, have not

been investigated in MEC pathogenesis at all, hence we have not suggested MTT for this targets.

In the end, we are left with 149 clinically applied drugs against asthma, inflammatory bowel diseases, different forms of arthritis, multiple sclerosis, cancer etc. (Additional file 2: Table S2). We have reduced the number of MTTs or drugs respectively down to a shorter list of top drugs (Fig. 5). This step is crucial, since the MTTs should be tested in an animal model of MEC first before applying for ethical approval for clinical studies, and screening of over a hundred drugs on thousands of animals is neither practical (with the realistic but complex MEC animal models), nor ethically sound.

The mandatory inclusion criteria for a drug is, that the corresponding pharmaceutical approach needs to be part of one of the described self-amplifying spirals of doom (Fig. 5). This is a vital point, since these self-amplifying mechanisms leading to the exaggerated inflammation is fundamental in MEC pathogenesis, epidermal proliferation and osteolysis.

From that group of drugs, all drugs approved for topical application or targeting at least two levels of a vicious circle simultaneously are included. Since the set of top drugs should include a drug for every approach shown in Table 1 we selected the drug not known to induce dermatological side effects, making them unsuitable for topic application to the middle ear or enhancing the likeliness of infection, something undesirable to have in the MEC diseased middle ear. NF- κ B has been repeatedly reported to contribute to the initial establishment as well as the fuelling and escalation of the inflammation and is a vital part of the three spirals

Target	Approach	Drug	First approval/ stage of trial	Application	Drug type	Ototoxic
TLR4	TLR4 expression	Ibudilast	Phase III	Oral	Small molecule	No
	MD-2-TLR4 antagonist	Taxanes	1995	Oral	Small molecule	No
		Ibudilast	Phase III	Oral	Small molecule	No
	TLR4-signalling	Naloxone	1971	Oral/topic	Small molecule	No
RAGE	RAGE antagonist	Azeliragon	Phase III	Oral	Small molecule	No
TLR4, RAGE, IL-1R, TNFR	BTK inhibitor	Fenebrutinib	Phase III	Oral	Small molecule	No
TLR4 and IL-1R	IRAK inhibitor	PF-06650833	Phase II	Oral	Small molecule	No
TLR4 and TNFR	c-IAP inhibitor	Birinapant	Phase II	Oral	Small molecule	No
TLR4, RAGE, TNFR	Pi3K inhibitor	Duvelisib	2018	Oral	Small molecule	No
Osteolysis/epidermal	Inflammasome	Tranilast	1982	Oral/topic	Small molecule	No
proliferation	Inhibitor	Dapansutrile	Phase II	Oral/topic	Small molecule	No
	IL-1R signalling	Diacerein	2008	Oral/topic	Small molecule	No
	IL-6R signalling	Upadacitinib	2019	Oral	Small molecule	No
	TNF-α signaling	IGURATIMOD	2012	Oral	Small molecule	No

Table 1 The 13 top drugs most recommended to be investigated in animal models of MEC

The top drugs were chosen in regard to a fast translation into clinic in accordance with the criteria described in the manuscript

of doom described, targeting this pathway should be most definitively considered. When careful in vivo evaluation unveiled a couple of MMT able to successfully target MEC, the complex interwoven inflammatory networks described in this review may offer synergistic interaction between these drugs. Hence we suggest investigating this by performing animal experiments enabling calculations of isobolograms. We also want to mention that the MTTs described in Additional file 2: Table S2 are limited due to our inclusion criteria.

Abbreviations

MEC: Middle ear cholesteatoma; MTT: Molecular targeted therapies; PAMP: Pathogen associated molecular pattern; DAMP: Damage associated molecular pattern; ECM: Extracellular matrix; LPS: Lipopolysaccharide; Mincle: Macrophage inducible Ca2+-dependent lectin receptor; TLR4: Toll-like receptor 4; RAGE: Receptor for advanced glycation end product; FRET: Förster resonance energy transfer; TREM-1: Triggering receptor expressed on myeloid cells-1; MIF: Macrophage migration inhibitory factor; HIF-1: Hypoxia-inducible factor 1; BTK: Bruton's Tyrosine Kinase; c-IAP: Cellular inhibitor of apoptosis proteins; HLA-DR: Human Leukocyte Antigen-DR isotype; RANKL: Receptor activator for nuclear factor kappa B ligand; PTHrP: Parathyroid-hormone-related protein; MMP: Matrix metalloproteinases; EGF: Epidermal growth factor; IL-1-RA: Interleukin-1 receptor antagonist.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s12964-022-00953-w.

Additional file 1: Supplementary table 1. Potential MEC targets which need further investigation regarding new drugs or their role.

Additional file 2: Supplementary table 2. All suggested drugs applicable in the middle ear context (light green: double targeting drugs / dark green: drugs double targeting on the same target / * = dizzienes or vertigo).

Author contributions

MS: established the conception and design of the article, interpreted the relevant literature and drafted the article. PG: substantially contributed to the design of the article and interpreted the relevant literature and revised it critically for important intellectual content. HS: contributed to the conception of the article and interpreted the relevant literature, revised it critically for important intellectual content. All authors read and approved the final manuscript.

Funding

Open Access funding enabled and organized by Projekt DEAL. This literature research was funded by the Federal Ministry of Education and Research.

Availability of data and materials

All data generated or analysed during this study are included in this published article [and its supplementary information files].

Declarations

Ethics approval and consent to participate

Not applicable.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Department of Otolaryngology, Head and Neck Surgery, Universität Bielefeld, Teutoburger Str. 50, 33604 Bielefeld, Germany. ²Department of Medicine, National University of Singapore, and National University Health System, Singapore, Singapore.

Received: 30 June 2022 Accepted: 5 August 2022 Published online: 19 September 2022

References

 Britze A, Moller ML, Ovesen T. Incidence, 10-year recidivism rate and prognostic factors for cholesteatoma. J Laryngol Otol. 2017;131(4):319–28.

- Kemppainen HO, Puhakka HJ, Laippala PJ, Sipila MM, Manninen MP, Karma PH. Epidemiology and aetiology of middle ear cholesteatoma. Acta Otolaryngol. 1999;119(5):568–72.
- Djurhuus BD, Skytthe A, Christensen K, Faber CE. Cholesteatoma in Danish children—a national study of changes in the incidence rate over 34 years. Int J Pediatr Otorhinolaryngol. 2015;79(2):127–30.
- Shibata S, Murakami K, Umeno Y, Komune S. Epidemiological study of cholesteatoma in Fukuoka City. J Laryngol Otol. 2015;129(Suppl 2):S6-11.
- Presutti L, Gioacchini FM, Alicandri-Ciufelli M, Villari D, Marchioni D. Results of endoscopic middle ear surgery for cholesteatoma treatment: a systematic review. Acta Otorhinolaryngol Ital. 2014;34(3):153–7.
- Moller PR, Pedersen CN, Grosfjeld LR, Faber CE, Djurhuus BD. Recurrence of cholesteatoma—a retrospective study including 1,006 patients for more than 33 years. Int Arch Otorhinolaryngol. 2020;24(1):e18–23.
- van der Toom HFE, van der Schroeff MP, Metselaar M, van Linge A, Vroegop JL, Pauw RJ. Treatment outcome of the bony obliteration tympanoplasty versus nonobliterative techniques in cholesteatoma surgery: a retrospective analysis. Otol Neurotol. 2021;42(9):1366–74.
- Morita Y, Takahashi K, Izumi S, Kubota Y, Ohshima S, Yamamoto Y, et al. Risk factors of recurrence in pediatric congenital cholesteatoma. Otol Neurotol. 2017;38(10):1463–9.
- 9. Kennedy KL, Singh AK. Middle ear cholesteatoma. Treasure Island: StatPearls; 2022.
- Sennaroglu L, Ozkul A, Gedikoglu G, Turan E. Effect of intratympanic steroid application on the development of experimental cholesteatoma. Laryngoscope. 1998;108(4 Pt 1):543–7.
- Kayhan FT, Algun Z. The effect of systemic prednisolone on propylene glycol-induced otitis media in guinea pig. Kulak Burun Bogaz Ihtis Derg. 2006;16(5):214–20.
- Prasad SC, Shin SH, Russo A, Di Trapani G, Sanna M. Current trends in the management of the complications of chronic otitis media with cholesteatoma. Curr Opin Otolaryngol Head Neck Surg. 2013;21(5):446–54.
- Ocal R, Kargin Kaytez S, Yumusak N, Akkoca O, Celik H, Arslan N. The effect of oral montelukast administration in cholesteatoma development and inflammation: an experimental animal model. Otol Neurotol. 2021;42(5):e568–72.
- Peek FA, Huisman MA, Berckmans RJ, Sturk A, Van Loon J, Grote JJ. Lipopolysaccharide concentration and bone resorption in cholesteatoma. Otol Neurotol. 2003;24(5):709–13.
- Sugita T, Huang CC, Abramson M. The effect of endotoxin and prostaglandin E2 on the proliferation of keratinocytes in vitro. Arch Otorhinolaryngol. 1986;243(3):211–4.
- Jung JY, Lee DH, Wang EW, Nason R, Sinnwell TM, Vogel JP, et al. *P. aeruginosa* infection increases morbidity in experimental cholesteatomas. Laryngoscope. 2011;121(11):2449–54.
- Likus W, Siemianowicz K, Markowski J, Wiaderkiewicz J, Kostrzab-Zdebel A, Jura-Szoltys E, et al. Bacterial infections and osteoclastogenesis regulators in men and women with cholesteatoma. Arch Immunol Ther Exp (Warsz). 2016;64(3):241–7.
- Jiang H, Wu C, Xu J, Wang Q, Shen L, Ou X, et al. Bacterial and fungal infections promote the bone erosion progression in acquired cholesteatoma revealed by metagenomic next-generation sequencing. Front Microbiol. 2021;12:761111.
- 19. Weiss JP, Antonelli PJ, Dirain CO. Microbiome analysis of cholesteatoma by gene sequencing. Otol Neurotol. 2019;40(9):1186–93.
- 20. Attallah MS. Microbiology of chronic suppurative otitis media with cholesteatoma. Saudi Med J. 2000;21(10):924–7.
- Effat KG, Madany NM. Mycological study on cholesteatoma keratin obtained during primary mastoid surgery. J Laryngol Otol. 2014;128(10):881–4.
- Chole RA, Hughes RM, Faddis BT. Keratin particle-induced osteolysis: a mouse model of inflammatory bone remodeling related to cholesteatoma. J Assoc Res Otolaryngol. 2001;2(1):65–71.
- lino Y, Toriyama M, Ohmi S, Kanegasaki S. Activation of peritoneal macrophages with human cholesteatoma debris and alpha-keratin. Acta Otolaryngol. 1990;109(5–6):444–9.
- Matino E, Venegas MP, Diez S, Pinart J. Antibiotic treatment in surgery for cholesteatoma. An Otorrinolaringol Ibero Am. 1999;26(5):445–55.

- 25. Ricciardiello F, Cavaliere M, Mesolella M, lengo M. Notes on the microbiology of cholesteatoma: clinical findings and treatment. Acta Otorhinolaryngol Ital. 2009;29(4):197–202.
- Pierce NE, Antonelli PJ. Efficacy of antibiotic prophylaxis prior to tympanoplasty for contaminated cholesteatoma. Laryngoscope. 2016;126(10):2363–6.
- Xu F, Kong W, Peng J, Gu H, Zheng H. Analysis of main pathogenic bacteria and drug sensitivity in patients with chronic suppurative otitis media and middle ear cholesteatoma in China. Biotechnol Lett. 2020;42(8):1559–66.
- Kazmierczak W, Janiak-Kiszka J, Budzynska A, Nowaczewska M, Kazmierczak H, Gospodarek-Komkowska E. Analysis of pathogens and antimicrobial treatment in different groups of patients with chronic otitis media. J Laryngol Otol. 2022;136(3):219–22.
- Vartiainen E. Factors associated with recurrence of cholesteatoma. J Laryngol Otol. 1995;109(7):590–2.
- Iino Y, İmamura Y, Kojima C, Takegoshi S, Suzuki JI. Risk factors for recurrent and residual cholesteatoma in children determined by second stage operation. Int J Pediatr Otorhinolaryngol. 1998;46(1–2):57–65.
- Szczepanski MJ, Luczak M, Olszewska E, Molinska-Glura M, Zagor M, Krzeski A, et al. Molecular signaling of the HMGB1/RAGE axis contributes to cholesteatoma pathogenesis. J Mol Med (Berl). 2015;93(3):305–14.
- Macias JD, Gerkin RD, Locke D, Macias MP. Differential gene expression in cholesteatoma by DNA chip analysis. Laryngoscope. 2013;123(Suppl S5):S1-21.
- Klenke C, Janowski S, Borck D, Widera D, Ebmeyer J, Kalinowski J, et al. Identification of novel cholesteatoma-related gene expression signatures using full-genome microarrays. PLoS ONE. 2012;7(12):e52718.
- Hirai H, Kariya S, Okano M, Fukushima K, Kataoka Y, Maeda Y, et al. Expression of toll-like receptors in chronic otitis media and cholesteatoma. Int J Pediatr Otorhinolaryngol. 2013;77(5):674–6.
- Si Y, Chen YB, Chen SJ, Zheng YQ, Liu X, Liu Y, et al. TLR4 drives the pathogenesis of acquired cholesteatoma by promoting local inflammation and bone destruction. Sci Rep. 2015;5:16683.
- Nagel J, Wollner S, Schurmann M, Brotzmann V, Muller J, Greiner JF, et al. Stem cells in middle ear cholesteatoma contribute to its pathogenesis. Sci Rep. 2018;8(1):6204.
- Jesic S, Jotic A, Tomanovic N, Zivkovic M. Expression of toll-like receptors 2, 4 and nuclear factor kappa B in mucosal lesions of human otitis: pattern and relationship in a clinical immunohistochemical study. Ann Otol Rhinol Laryngol. 2014;123(6):434–41.
- Lee HY, Park MS, Byun JY, Kim YI, Yeo SG. Expression of pattern recognition receptors in cholesteatoma. Eur Arch Otorhinolaryngol. 2014;271(2):245–53.
- Ho KY, Yeh TS, Huang HH, Hung KF, Chai CY, Chen WT, et al. Upregulation of phosphorylated HSP27, PRDX2, GRP75, GRP78 and GRP94 in acquired middle ear cholesteatoma growth. Int J Mol Sci. 2013;14(7):14439–59.
- 40. Shinoda H, Huang CC. Heat shock proteins in middle ear cholesteatoma. Otolaryngol Head Neck Surg. 1996;114(1):77–83.
- Triantafilou M, Triantafilou K. Heat-shock protein 70 and heat-shock protein 90 associate with Toll-like receptor 4 in response to bacterial lipopolysaccharide. Biochem Soc Trans. 2004;32(Pt 4):636–9.
- Schurmann M, Greiner JFW, Volland-Thurn V, Oppel F, Kaltschmidt C, Sudhoff H, et al. Stem cell-induced inflammation in cholesteatoma is inhibited by the TLR4 antagonist LPS-RS. Cells. 2020;9(1):199.
- Schurmann M, Oppel F, Shao S, Volland-Thurn V, Kaltschmidt C, Kaltschmidt B, et al. Chronic inflammation of middle ear cholesteatoma promotes its recurrence via a paracrine mechanism. Cell Commun Signal. 2021;19(1):25.
- Kim MG, Park DC, Oh IH, Kim YI, Choi SA, Jung SY, et al. Increased expression of Dec-205, Bcl-10, Tim-3, and Trem-1 mRNA in chronic otitis media with cholesteatoma. Acta Otolaryngol. 2014;134(5):475–80.
- Tammaro A, Derive M, Gibot S, Leemans JC, Florquin S, Dessing MC. TREM-1 and its potential ligands in non-infectious diseases: from biology to clinical perspectives. Pharmacol Ther. 2017;177:81–95.
- Roger T, David J, Glauser MP, Calandra T. MIF regulates innate immune responses through modulation of Toll-like receptor 4. Nature. 2001;414(6866):920–4.

- 47. Choufani G, Ghanooni R, Decaestecker C, Delbrouck K, Simon P, Schuring MP, et al. Detection of macrophage migration inhibitory factor (MIF) in human cholesteatomas and functional implications of correlations to recurrence status and to expression of matrix metalloproteinases-3/9, retinoic acid receptor-beta, and anti-apoptotic galectin-3. Laryngoscope. 2001;111(9):1656–62.
- 48. Qin YH, Dai SM, Tang GS, Zhang J, Ren D, Wang ZW, et al. HMGB1 enhances the proinflammatory activity of lipopolysaccharide by promoting the phosphorylation of MAPK p38 through receptor for advanced glycation end products. J Immunol. 2009;183(10):6244–50.
- Kim SH, Han SH, Byun JY, Park MS, Kim YI, Yeo SG. Expression of C-type lectin receptor mRNA in chronic otitis media with cholesteatoma. Acta Otolaryngol. 2017;137(6):581–7.
- Dejban P, Sahraei M, Chamanara M, Dehpour A, Rashidian A. Antiinflammatory effect of amitriptyline in a rat model of acetic acidinduced colitis: the involvement of the TLR4/NF-kB signaling pathway. Fundam Clin Pharmacol. 2021;35(5):843–51.
- Cirillo PF, Asojo OA, Khire U, Lee Y, Mootien S, Hegan P, et al. Inhibition of macrophage migration inhibitory factor by a chimera of two allosteric binders. ACS Med Chem Lett. 2020;11(10):1843–7.
- Fox RJ, Coffey CS, Conwit R, Cudkowicz ME, Gleason T, Goodman A, et al. Phase 2 trial of ibudilast in progressive multiple sclerosis. N Engl J Med. 2018;379(9):846–55.
- Meza-Romero R, Benedek G, Leng L, Bucala R, Vandenbark AA. Predicted structure of MIF/CD74 and RTL1000/CD74 complexes. Metab Brain Dis. 2016;31(2):249–55.
- Offner H, Sinha S, Burrows GG, Ferro AJ, Vandenbark AA. RTL therapy for multiple sclerosis: a Phase I clinical study. J Neuroimmunol. 2011;231(1–2):7–14.
- Yadav V, Bourdette DN, Bowen JD, Lynch SG, Mattson D, Preiningerova J, et al. Recombinant T-Cell Receptor Ligand (RTL) for treatment of multiple sclerosis: a double-blind, placebo-controlled, Phase 1, doseescalation study. Autoimmune Dis. 2012;2012:954739.
- Tidswell M, Tillis W, Larosa SP, Lynn M, Wittek AE, Kao R, et al. Phase 2 trial of eritoran tetrasodium (E5564), a toll-like receptor 4 antagonist, in patients with severe sepsis. Crit Care Med. 2010;38(1):72–83.
- 57. Opal SM, Laterre PF, Francois B, LaRosa SP, Angus DC, Mira JP, et al. Effect of eritoran, an antagonist of MD2-TLR4, on mortality in patients with severe sepsis: the ACCESS randomized trial. JAMA. 2013;309(11):1154–62.
- Sundaram V, Morgan TR. Will studies in nonalcoholic steatohepatitis help manage alcoholic steatohepatitis? Clin Liver Dis. 2019;23(1):157–65.
- White JR, Phillips F, Monaghan T, Fateen W, Samuel S, Ghosh S, et al. Review article: novel oral-targeted therapies in inflammatory bowel disease. Aliment Pharmacol Ther. 2018;47(12):1610–22.
- Fox RJ, Raska P, Barro C, Karafa M, Konig V, Bermel RA, et al. Neurofilament light chain in a phase 2 clinical trial of ibudilast in progressive multiple sclerosis. Mult Scler. 2021;27(13):2014–22.
- de Grooth GJ, Kuivenhoven JA, Stalenhoef AF, de Graaf J, Zwinderman AH, Posma JL, et al. Efficacy and safety of a novel cholesteryl ester transfer protein inhibitor, JTT-705, in humans: a randomized phase II dose-response study. Circulation. 2002;105(18):2159–65.
- Razavi BM, Ghasemzadeh Rahbardar M, Hosseinzadeh H. A review of therapeutic potentials of turmeric (*Curcuma longa*) and its active constituent, curcumin, on inflammatory disorders, pain, and their related patents. Phytother Res. 2021;35(12):6489–513.
- Resman N, Gradisar H, Vasl J, Keber MM, Pristovsek P, Jerala R. Taxanes inhibit human TLR4 signaling by binding to MD-2. FEBS Lett. 2008;582(28):3929–34.
- Cella D, Peterman A, Hudgens S, Webster K, Socinski MA. Measuring the side effects of taxane therapy in oncology: the functional assessment of cancer therapy-taxane (FACT-taxane). Cancer. 2003;98(4):822–31.
- Rice TW, Wheeler AP, Bernard GR, Vincent JL, Angus DC, Aikawa N, et al. A randomized, double-blind, placebo-controlled trial of TAK-242 for the treatment of severe sepsis. Crit Care Med. 2010;38(8):1685–94.
- 66. li M, Matsunaga N, Hazeki K, Nakamura K, Takashima K, Seya T, et al. A novel cyclohexene derivative, ethyl (6R)-6-[N-(2-Chloro-4-fluorophenyl) sulfamoyl]cyclohex-1-ene-1-carboxylate (TAK-242), selectively inhibits toll-like receptor 4-mediated cytokine production through suppression of intracellular signaling. Mol Pharmacol. 2006;69(4):1288–95.

- 67. Khojah HM, Ahmed S, Abdel-Rahman MS, Elhakeim EH. Resveratrol as an effective adjuvant therapy in the management of rheumatoid arthritis: a clinical study. Clin Rheumatol. 2018;37(8):2035–42.
- Kaku K, Sumino S, Katou M, Nishiyama Y, Kinugawa Y. Randomized, double-blind, phase III study to evaluate the efficacy and safety of once-daily treatment with alogliptin and metformin hydrochloride in Japanese patients with type 2 diabetes. Diabetes Obes Metab. 2017;19(3):463–7.
- Jakus PB, Kalman N, Antus C, Radnai B, Tucsek Z, Gallyas F Jr, et al. TRAF6 is functional in inhibition of TLR4-mediated NF-kappaB activation by resveratrol. J Nutr Biochem. 2013;24(5):819–23.
- Ta NN, Li Y, Schuyler CA, Lopes-Virella MF, Huang Y. DPP-4 (CD26) inhibitor alogliptin inhibits TLR4-mediated ERK activation and ERKdependent MMP-1 expression by U937 histiocytes. Atherosclerosis. 2010;213(2):429–35.
- Wang X, Zhang Y, Peng Y, Hutchinson MR, Rice KC, Yin H, et al. Pharmacological characterization of the opioid inactive isomers (+)-naltrexone and (+)-naloxone as antagonists of toll-like receptor 4. Br J Pharmacol. 2016;173(5):856–69.
- Yuan Z, Syed MA, Panchal D, Rogers D, Joo M, Sadikot RT. Curcumin mediated epigenetic modulation inhibits TREM-1 expression in response to lipopolysaccharide. Int J Biochem Cell Biol. 2012;44(11):2032–43.
- Cuvier V, Lorch U, Witte S, Olivier A, Gibot S, Delor I, et al. A first-in-man safety and pharmacokinetics study of nangibotide, a new modulator of innate immune response through TREM-1 receptor inhibition. Br J Clin Pharmacol. 2018;84(10):2270–9.
- Francois B, Wittebole X, Ferrer R, Mira JP, Dugernier T, Gibot S, et al. Nangibotide in patients with septic shock: a Phase 2a randomized controlled clinical trial. Intensive Care Med. 2020;46(7):1425–37.
- Francois B, Lambden S, Gibot S, Derive M, Olivier A, Cuvier V, et al. Rationale and protocol for the efficacy, safety and tolerability of nangibotide in patients with septic shock (ASTONISH) phase IIb randomised controlled trial. BMJ Open. 2021;11(7):e042921.
- Burstein AH, Grimes I, Galasko DR, Aisen PS, Sabbagh M, Mjalli AM. Effect of TTP488 in patients with mild to moderate Alzheimer's disease. BMC Neurol. 2014;14:12.
- 77. Li Y, Yao J, Han C, Yang J, Chaudhry MT, Wang S, et al. Quercetin, inflammation and immunity. Nutrients. 2016;8(3):167.
- Tan RZ, Liu J, Zhang YY, Wang HL, Li JC, Liu YH, et al. Curcumin relieved cisplatin-induced kidney inflammation through inhibiting Mincle-maintained M1 macrophage phenotype. Phytomedicine. 2019;52:284–94.
- Mohammadi A, Blesso CN, Barreto GE, Banach M, Majeed M, Sahebkar A. Macrophage plasticity, polarization and function in response to curcumin, a diet-derived polyphenol, as an immunomodulatory agent. J Nutr Biochem. 2019;66:1–16.
- Kuczkowski J, Sakowicz-Burkiewicz M, Izycka-Swieszewska E, Mikaszewski B, Pawelczyk T. Expression of tumor necrosis factor-alpha, interleukin-1alpha, interleukin-6 and interleukin-10 in chronic otitis media with bone osteolysis. ORL J Otorhinolaryngol Relat Spec. 2011;73(2):93–9.
- Imai R, Sato T, Iwamoto Y, Hanada Y, Terao M, Ohta Y, et al. Osteoclasts modulate bone erosion in cholesteatoma via RANKL signaling. J Assoc Res Otolaryngol. 2019;20(5):449–59.
- Bujia J, Kim C, Ostos-Aumente P, Lopez-Villarejo J, Kastenbauer E. Enhanced epithelial proliferation due to elevated levels of interleukin-1 receptors in middle ear cholesteatomas. Eur Arch Otorhinolaryngol. 1997;254(1):6–8.
- Xu Y, Tao ZZ, Hua QQ, Wang XX, Xiao BK. Expression and activation of nuclear factor-kappaB in middle ear cholesteatoma. Zhonghua Er Bi Yan Hou Tou Jing Wai Ke Za Zhi. 2006;41(6):455–9.
- Dambergs K, Sumeraga G, Pilmane M. Complex evaluation of tissue factors in pediatric cholesteatoma. Children (Basel). 2021;8(10):926.
- Zhao Y, Ling F, Wang HC, Sun XH. Chronic TLR signaling impairs the long-term repopulating potential of hematopoietic stem cells of wild type but not Id1 deficient mice. PLoS ONE. 2013;8(2):e55552.
- Zhang QA, Hamajima Y, Zhang Q, Lin J. Identification of Id1 in acquired middle ear cholesteatoma. Arch Otolaryngol Head Neck Surg. 2008;134(3):306–10.
- 87. Hamajima Y, Komori M, Preciado DA, Choo DI, Moribe K, Murakami S, et al. The role of inhibitor of DNA-binding (ld1) in hyperproliferation

of keratinocytes: the pathological basis for middle ear cholesteatoma from chronic otitis media. Cell Prolif. 2010;43(5):457–63.

- Huang CC, Chen CT, Huang TS, Shinoda H. Mediation of signal transduction in keratinocytes of human middle ear cholesteatoma by ras protein. Eur Arch Otorhinolaryngol. 1996;253(7):385–9.
- Shinoda H, Huang CC. Expressions of c-jun and p53 proteins in human middle ear cholesteatoma: relationship to keratinocyte proliferation, differentiation, and programmed cell death. Laryngoscope. 1995;105(11):1232–7.
- Adunka O, Gstoettner W, Knecht R, Kierner AC. Expression of hypoxia inducible factor 1 alpha and Von Hippel Lindau protein in human middle ear cholesteatoma. Laryngoscope. 2003;113(7):1210–5.
- Zhang C, Chen M, Tao Q, Chi Z. Cobalt chloride-stimulated hypoxia promotes the proliferation of cholesteatoma keratinocytes via the PI3K/ Akt signaling pathway. Int J Med Sci. 2021;18(15):3403–11.
- 92. Ramadass V, Vaiyapuri T, Tergaonkar V. Small molecule NF-kappaB pathway inhibitors in clinic. Int J Mol Sci. 2020;21(14):5164.
- Markham A, Dhillon S. Acalabrutinib: first global approval. Drugs. 2018;78(1):139–45.
- Byrd JC, Hillmen P, Ghia P, Kater AP, Chanan-Khan A, Furman RR, et al. Acalabrutinib versus ibrutinib in previously treated chronic lymphocytic leukemia: results of the first randomized phase III trial. J Clin Oncol. 2021;39(31):3441–52.
- 95. Syed YY. Zanubrutinib: first approval. Drugs. 2020;80(1):91-7.
- Narita Y, Nagane M, Mishima K, Terui Y, Arakawa Y, Yonezawa H, et al. Phase I/II study of tirabrutinib, a second-generation Bruton's tyrosine kinase inhibitor, in relapsed/refractory primary central nervous system lymphoma. Neuro Oncol. 2021;23(1):122–33.
- Mendes-Bastos P, Brasileiro A, Kolkhir P, Frischbutter S, Scheffel J, Monino-Romero S, et al. Bruton's tyrosine kinase inhibition—an emerging therapeutic strategy in immune-mediated dermatological conditions. Allergy. 2022. https://doi.org/10.1111/all.15261.
- Winkler A, Sun W, De S, Jiao A, Sharif MN, Symanowicz PT, et al. The interleukin-1 receptor-associated kinase 4 inhibitor PF-06650833 blocks inflammation in preclinical models of rheumatic disease and in humans enrolled in a randomized clinical trial. Arthritis Rheumatol. 2021;73(12):2206–18.
- Amaravadi RK, Schilder RJ, Martin LP, Levin M, Graham MA, Weng DE, et al. A Phase I study of the SMAC-mimetic birinapant in adults with refractory solid tumors or lymphoma. Mol Cancer Ther. 2015;14(11):2569–75.
- Miller BW, Przepiorka D, de Claro RA, Lee K, Nie L, Simpson N, et al. FDA approval: idelalisib monotherapy for the treatment of patients with follicular lymphoma and small lymphocytic lymphoma. Clin Cancer Res. 2015;21(7):1525–9.
- 101. Markham A. Copanlisib: first global approval. Drugs. 2017;77(18):2057–62.
- Rodrigues DA, Sagrillo FS, Fraga CAM. Duvelisib: a 2018 novel FDAapproved small molecule inhibiting phosphoinositide 3-kinases. Pharmaceuticals (Basel). 2019;12(2):69.
- Narayan P, Prowell TM, Gao JJ, Fernandes LL, Li E, Jiang X, et al. FDA approval summary: alpelisib plus fulvestrant for patients with HRpositive, HER2-negative, PIK3CA-mutated, advanced or metastatic breast cancer. Clin Cancer Res. 2021;27(7):1842–9.
- Hovstadius P, Larsson R, Jonsson E, Skov T, Kissmeyer AM, Krasilnikoff K, et al. A Phase I study of CHS 828 in patients with solid tumor malignancy. Clin Cancer Res. 2002;8(9):2843–50.
- Lee JC, Menacherry S, Diehl MC, Giffear MD, White CJ, Juba R, et al. Safety, bioavailability, and pharmacokinetics of VGX-1027-A novel oral anti-inflammatory drug in healthy human subjects. Clin Pharmacol Drug Dev. 2016;5(2):91–101.
- 106. Suh JJ, Pettinati HM, Kampman KM, O'Brien CP. The status of disulfiram: a half of a century later. J Clin Psychopharmacol. 2006;26(3):290–302.
- 107. Shirley M. Ixazomib: first global approval. Drugs. 2016;76(3):405–11.
- Wang Z, Yang J, Kirk C, Fang Y, Alsina M, Badros A, et al. Clinical pharmacokinetics, metabolism, and drug–drug interaction of carfilzomib. Drug Metab Dispos. 2013;41(1):230–7.
- 109. Ghobrial IM, Vij R, Siegel D, Badros A, Kaufman J, Raje N, et al. A Phase Ib/II study of oprozomib in patients with advanced multiple myeloma and waldenstrom macroglobulinemia. Clin Cancer Res. 2019;25(16):4907–16.

- 110. Spencer A, Harrison S, Zonder J, Badros A, Laubach J, Bergin K, et al. A phase 1 clinical trial evaluating marizomib, pomalidomide and lowdose dexamethasone in relapsed and refractory multiple myeloma (NPI-0052-107): final study results. Br J Haematol. 2018;180(1):41–51.
- 111. Fallah J, Rini Bl. HIF inhibitors: status of current clinical development. Curr Oncol Rep. 2019;21(1):6.
- 112. Sun X, Lv X, Yan Y, Zhao Y, Ma R, He M, et al. Hypoxia-mediated cancer stem cell resistance and targeted therapy. Biomed Pharmacother. 2020;130:110623.
- Ibrahim N, Buchbinder EI, Granter SR, Rodig SJ, Giobbie-Hurder A, Becerra C, et al. A phase I trial of panobinostat (LBH589) in patients with metastatic melanoma. Cancer Med. 2016;5(11):3041–50.
- Haas NB, Quirt I, Hotte S, McWhirter E, Polintan R, Litwin S, et al. Phase Il trial of vorinostat in advanced melanoma. Investig New Drugs. 2014;32(3):526–34.
- 115. Nowakowski GS, McCollum AK, Ames MM, Mandrekar SJ, Reid JM, Adjei AA, et al. A phase I trial of twice-weekly 17-allylamino-demethoxy-geldanamycin in patients with advanced cancer. Clin Cancer Res. 2006;12(20 Pt 1):6087–93.
- Oki Y, Copeland A, Romaguera J, Fayad L, Fanale M, Faria Sde C, et al. Clinical experience with the heat shock protein-90 inhibitor, tanespimycin, in patients with relapsed lymphoma. Leuk Lymphoma. 2012;53(5):990–2.
- 117. Popat R, Plesner T, Davies F, Cook G, Cook M, Elliott P, et al. A phase 2 study of SRT501 (resveratrol) with bortezomib for patients with relapsed and or refractory multiple myeloma. Br J Haematol. 2013;160(5):714–7.
- Garrett CR, Bekaii-Saab TS, Ryan T, Fisher GA, Clive S, Kavan P, et al. Randomized phase 2 study of pegylated SN-38 (EZN-2208) or irinotecan plus cetuximab in patients with advanced colorectal cancer. Cancer. 2013;119(24):4223–30.
- 119. Jeong W, Rapisarda A, Park SR, Kinders RJ, Chen A, Melillo G, et al. Pilot trial of EZN-2968, an antisense oligonucleotide inhibitor of hypoxiainducible factor-1 alpha (HIF-1alpha), in patients with refractory solid tumors. Cancer Chemother Pharmacol. 2014;73(2):343–8.
- 120. Bruce JY, Eickhoff J, Pili R, Logan T, Carducci M, Arnott J, et al. A phase II study of 2-methoxyestradiol nanocrystal colloidal dispersion alone and in combination with sunitinib malate in patients with metastatic renal cell carcinoma progressing on sunitinib malate. Investig New Drugs. 2012;30(2):794–802.
- 121. Gradishar WJ, Vogelzang NJ, Kilton LJ, Leibach SJ, Rademaker AW, French S, et al. A phase II clinical trial of echinomycin in metastatic soft tissue sarcoma. An Illinois Cancer Center Study. Investig New Drugs. 1995;13(2):171–4.
- 122. Lopez L, Velez I, Asela C, Cruz C, Alves F, Robledo S, et al. A phase II study to evaluate the safety and efficacy of topical 3% amphotericin B cream (Anfoleish) for the treatment of uncomplicated cutaneous leishmaniasis in Colombia. PLoS Negl Trop Dis. 2018;12(7):e0006653.
- Nehme R, Hallal R, El Dor M, Kobeissy F, Gouilleux F, Mazurier F, et al. Repurposing of acriflavine to target chronic myeloid leukemia treatment. Curr Med Chem. 2021;28(11):2218–33.
- 124. Zhao Z, Bo Z, Gong W, Guo Y. Inhibitor of differentiation 1 (ld1) in cancer and cancer therapy. Int J Med Sci. 2020;17(8):995–1005.
- 125. Takahashi H, Kawaguchi M, Kitamura K, Narumiya S, Kawamura M, Tengan I, et al. An exploratory study on the anti-inflammatory effects of fucoidan in relation to quality of life in advanced cancer patients. Integr Cancer Ther. 2018;17(2):282–91.
- 126. Ye Y, Liu X, Wu N, Han Y, Wang J, Yu Y, et al. Efficacy and safety of berberine alone for several metabolic disorders: a systematic review and meta-analysis of randomized clinical trials. Front Pharmacol. 2021;12:653887.
- 127. Hussein MR, Sayed RH, Abu-Dief EE. Immune cell profile in invasive cholesteatomas: preliminary findings. Exp Mol Pathol. 2010;88(2):316–23.
- Negri B, Schilling V, Bujia J, Schulz P, Kastenbauer E. Immunotype findings in macrophages in aural cholesteatomas. Eur Arch Otorhinolaryngol. 1992;249(2):87–90.
- 129. Zhang E, Gong S. Expression and significance of HIF-1alpha and iNOS in human middle ear cholesteatoma epithelium. Lin Chung Er Bi Yan Hou Tou Jing Wai Ke Za Zhi. 2007;21(10):463–5.
- Ferlito A, Devaney KO, Rinaldo A, Milroy CM, Wenig BM, Iurato S, et al. Clinicopathological consultation. Ear cholesteatoma versus cholesterol granuloma. Ann Otol Rhinol Laryngol. 1997;106(1):79–85.

- Jiang H, Si Y, Li Z, Huang X, Chen S, Zheng Y, et al. TREM-2 promotes acquired cholesteatoma-induced bone destruction by modulating TLR4 signaling pathway and osteoclasts activation. Sci Rep. 2016;6:38761.
- Jin S, Meng C, He Y, Wang X, Zhang Q, Wang Z, et al. Curcumin prevents osteocyte apoptosis by inhibiting M1-type macrophage polarization in mice model of glucocorticoid-associated osteonecrosis of the femoral head. J Orthop Res. 2020;38(9):2020–30.
- 133. Zhang H, Cao N, Yang Z, Fang X, Yang X, Li H, et al. Bilobalide alleviated dextran sulfate sodium-induced experimental colitis by inhibiting M1 macrophage polarization through the NF-kappaB signaling pathway. Front Pharmacol. 2020;11:718.
- Fan H, Tang HB, Shan LQ, Liu SC, Huang DG, Chen X, et al. Quercetin prevents necroptosis of oligodendrocytes by inhibiting macrophages/ microglia polarization to M1 phenotype after spinal cord injury in rats. J Neuroinflamm. 2019;16(1):206.
- Lu H, Wu L, Liu L, Ruan Q, Zhang X, Hong W, et al. Quercetin ameliorates kidney injury and fibrosis by modulating M1/M2 macrophage polarization. Biochem Pharmacol. 2018;154:203–12.
- Liu Y, Liu X, Hua W, Wei Q, Fang X, Zhao Z, et al. Berberine inhibits macrophage M1 polarization via AKT1/SOCS1/NF-kappaB signaling pathway to protect against DSS-induced colitis. Int Immunopharmacol. 2018;57:121–31.
- Feng H, Yin Y, Zheng R, Kang J. Rosiglitazone ameliorated airway inflammation induced by cigarette smoke via inhibiting the M1 macrophage polarization by activating PPARgamma and RXRalpha. Int Immunopharmacol. 2021;97:107809.
- Hyam SR, Lee IA, Gu W, Kim KA, Jeong JJ, Jang SE, et al. Arctigenin ameliorates inflammation in vitro and in vivo by inhibiting the PI3K/AKT pathway and polarizing M1 macrophages to M2-like macrophages. Eur J Pharmacol. 2013;708(1–3):21–9.
- Ali MA, Thrower SL, Hanna SJ, Coulman SA, Birchall JC, Wong FS, et al. Topical steroid therapy induces pro-tolerogenic changes in Langerhans cells in human skin. Immunology. 2015;146(3):411–22.
- Navarro-Barriuso J, Mansilla MJ, Naranjo-Gomez M, Sanchez-Pla A, Quirant-Sanchez B, Teniente-Serra A, et al. Comparative transcriptomic profile of tolerogenic dendritic cells differentiated with vitamin D3, dexamethasone and rapamycin. Sci Rep. 2018;8(1):14985.
- 141. Moreau A, Alliot-Licht B, Cuturi MC, Blancho G. Tolerogenic dendritic cell therapy in organ transplantation. Transpl Int. 2017;30(8):754–64.
- 142. Metz M, Grimbaldeston MA, Nakae S, Piliponsky AM, Tsai M, Galli SJ. Mast cells in the promotion and limitation of chronic inflammation. Immunol Rev. 2007;217:304–28.
- 143. Albino AP, Reed JA, Bogdany JK, Sassoon J, Parisier SC. Increased numbers of mast cells in human middle ear cholesteatomas: implications for treatment. Am J Otol. 1998;19(3):266–72.
- Ottaviani F, Neglia CB, Berti E. Cytokines and adhesion molecules in middle ear cholesteatoma. A role in epithelial growth? Acta Otolaryngol. 1999;119(4):462–7.
- Berger G, Hawke M, Ekem JK. Bone resorption in chronic otitis media. The role of mast cells. Acta Otolaryngol. 1985;100(1–2):72–80.
- Ragipoglu D, Dudeck A, Haffner-Luntzer M, Voss M, Kroner J, Ignatius A, et al. The role of mast cells in bone metabolism and bone disorders. Front Immunol. 2020;11:163.
- Branco A, Yoshikawa FSY, Pietrobon AJ, Sato MN. Role of histamine in modulating the immune response and inflammation. Mediat Inflamm. 2018;2018:9524075.
- 148. Zhang T, Finn DF, Barlow JW, Walsh JJ. Mast cell stabilisers. Eur J Pharmacol. 2016;778:158–68.
- 149. Motola D, Donati M, Biagi C, Calamelli E, Cipriani F, Melis M, et al. Safety profile of H1-antihistamines in pediatrics: an analysis based on data from VigiBase. Pharmacoepidemiol Drug Saf. 2017;26(10):1164–71.
- 150. Church MK, Church DS. Pharmacology of antihistamines. Indian J Dermatol. 2013;58(3):219–24.
- Cataldi M, Borriello F, Granata F, Annunziato L, Marone G. Histamine receptors and antihistamines: from discovery to clinical applications. Chem Immunol Allergy. 2014;100:214–26.
- 152. Chung BY, Kim HB, Jung MJ, Kang SY, Kwak IS, Park CW, et al. Post-burn pruritus. Int J Mol Sci. 2020;21(11):3880.
- 153. Thurmond RL. The histamine H4 receptor: from orphan to the clinic. Front Pharmacol. 2015;6:65.

- 154. Thangam EB, Jemima EA, Singh H, Baig MS, Khan M, Mathias CB, et al. The role of histamine and histamine receptors in mast cell-mediated allergy and inflammation: the hunt for new therapeutic targets. Front Immunol. 2018;9:1873.
- 155. Frankel E, Song M, Li S, Jiang J, Thurmond RL, Randazzo B. Efficacy and safety of toreforant, a selective histamine H4 receptor antagonist, for the treatment of moderate-to-severe plaque psoriasis: results from a phase 2 multicenter, randomized, double-blind, placebo-controlled trial. J Drugs Dermatol. 2018;17(8):873–9.
- 156. Murata Y, Song M, Kikuchi H, Hisamichi K, Xu XL, Greenspan A, et al. Phase 2a, randomized, double-blind, placebo-controlled, multicenter, parallel-group study of a H4 R-antagonist (JNJ-39758979) in Japanese adults with moderate atopic dermatitis. J Dermatol. 2015;42(2):129–39.
- 157. Werfel T, Layton G, Yeadon M, Whitlock L, Osterloh I, Jimenez P, et al. Efficacy and safety of the histamine H4 receptor antagonist ZPL-3893787 in patients with atopic dermatitis. J Allergy Clin Immunol. 2019;143(5):1830-7.e4.
- 158. Akimoto R, Pawankar R, Yagi T, Baba S. Acquired and congenital cholesteatoma: determination of tumor necrosis factor-alpha, intercellular adhesion molecule-1, interleukin-1-alpha and lymphocyte functional antigen-1 in the inflammatory process. ORL J Otorhinolaryngol Relat Spec. 2000;62(5):257–65.
- Schilling V, Bujia J, Negri B, Schulz P, Kastenbauer E. Immunologically activated cells in aural cholesteatoma. Am J Otolaryngol. 1991;12(5):249–53.
- Panina-Bordignon P, Mazzeo D, Lucia PD, D'Ambrosio D, Lang R, Fabbri L, et al. Beta2-agonists prevent Th1 development by selective inhibition of interleukin 12. J Clin Investig. 1997;100(6):1513–9.
- Agarwal SK, Marshall GD Jr. Beta-adrenergic modulation of human type-1/type-2 cytokine balance. J Allergy Clin Immunol. 2000;105(1 Pt 1):91–8.
- Cazzola M, Calzetta L, Matera MG. beta(2) -adrenoceptor agonists: current and future direction. Br J Pharmacol. 2011;163(1):4–17.
- Miyaura H, Iwata M. Direct and indirect inhibition of Th1 development by progesterone and glucocorticoids. J Immunol. 2002;168(3):1087–94.
- Franchimont D, Galon J, Gadina M, Visconti R, Zhou Y, Aringer M, et al. Inhibition of Th1 immune response by glucocorticoids: dexamethasone selectively inhibits IL-12-induced Stat4 phosphorylation in T lymphocytes. J Immunol. 2000;164(4):1768–74.
- 165. Kong W, Hooper KM, Ganea D. The natural dual cyclooxygenase and 5-lipoxygenase inhibitor flavocoxid is protective in EAE through effects on Th1/Th17 differentiation and macrophage/microglia activation. Brain Behav Immun. 2016;53:59–71.
- 166. Bing X, Xuelei L, Wanwei D, Linlang L, Keyan C. EGCG maintains Th1/ Th2 balance and mitigates ulcerative colitis induced by dextran sulfate sodium through TLR4/MyD88/NF-kappaB signaling pathway in rats. Can J Gastroenterol Hepatol. 2017;2017:3057268.
- Bielekova B, Lincoln A, McFarland H, Martin R. Therapeutic potential of phosphodiesterase-4 and -3 inhibitors in Th1-mediated autoimmune diseases. J Immunol. 2000;164(2):1117–24.
- Heystek HC, Thierry AC, Soulard P, Moulon C. Phosphodiesterase 4 inhibitors reduce human dendritic cell inflammatory cytokine production and Th1-polarizing capacity. Int Immunol. 2003;15(7):827–35.
- 169. Gianello V, Salvi V, Parola C, Moretto N, Facchinetti F, Civelli M, et al. The PDE4 inhibitor CHF6001 modulates pro-inflammatory cytokines, chemokines and Th1- and Th17-polarizing cytokines in human dendritic cells. Biochem Pharmacol. 2019;163:371–80.
- 170. Lemire JM, Archer DC, Beck L, Spiegelberg HL. Immunosuppressive actions of 1,25-dihydroxyvitamin D3: preferential inhibition of Th1 functions. J Nutr. 1995;125(6 Suppl):1704S-S1708.
- Bujia J, Holly A, Kim C, Scanady N, Kastenbauer E. Expression of human intercellular adhesion molecules in middle ear cholesteatoma. Am J Otolaryngol. 1994;15(4):271–5.
- 172. Shinoda H, Huang CC. Localization of intercellular adhesion molecule-1 in middle ear cholesteatoma. Eur Arch Otorhinolaryngol. 1995;252(7):385–90.
- Sharp C, Warren A, Oshima T, Williams L, Li JH, Alexander JS. Poly ADP ribose-polymerase inhibitors prevent the upregulation of ICAM-1 and E-selectin in response to Th1 cytokine stimulation. Inflammation. 2001;25(3):157–63.

- 174. Yetiser S, Satar B, Aydin N. Expression of epidermal growth factor, tumor necrosis factor-alpha, and interleukin-1alpha in chronic otitis media with or without cholesteatoma. Otol Neurotol. 2002;23(5):647–52.
- 175. Zhang Y, Liu H, Tang W, Qiu Q, Peng J. Resveratrol prevents TNFalpha-induced VCAM-1 and ICAM-1 upregulation in endothelial progenitor cells via reduction of NF-kappaB activation. J Int Med Res. 2020;48(9):300060520945131.
- Rajitha P, Biswas R, Sabitha M, Jayakumar R. Methotrexate in the treatment of psoriasis and rheumatoid arthritis: mechanistic insights, current issues and novel delivery approaches. Curr Pharm Des. 2017;23(24):3550–66.
- 177. Alidadi M, Liberale L, Montecucco F, Majeed M, Al-Rasadi K, Banach M, et al. Protective effects of curcumin on endothelium: an updated review. Adv Exp Med Biol. 2021;1291:103–19.
- Wiendl M, Becker E, Muller TM, Voskens CJ, Neurath MF, Zundler S. Targeting immune cell trafficking—insights from research models and implications for future IBD therapy. Front Immunol. 2021;12:656452.
- 179. Szczepanski M, Szyfter W, Jenek R, Wrobel M, Lisewska IM, Zeromski J. Toll-like receptors 2, 3 and 4 (TLR-2, TLR-3 and TLR-4) are expressed in the microenvironment of human acquired cholesteatoma. Eur Arch Otorhinolaryngol. 2006;263(7):603–7.
- Fang L, Chen L, Lin B, Han L, Zhu K, Song Q. Analysis of inflammatory and homeostatic roles of tissue-resident macrophages in the progression of cholesteatoma by RNA-Seq. Immunol Investig. 2021;50(6):609–21.
- Negri R, Schilling V, Bujia J, Kastenbauer E. Immunohistochemical identification of cholesteatoma-associated macrophage populations. Laryngorhinootologie. 1992;71(3):168–73.
- 182. Schink M, Troger W, Dabidian A, Goyert A, Scheuerecker H, Meyer J, et al. Mistletoe extract reduces the surgical suppression of natural killer cell activity in cancer patients. A randomized phase III trial. Forsch Komplementmed. 2007;14(1):9–17.
- Jesic SD, Jotic AD, Babic BB. Predictors for sensorineural hearing loss in patients with tubotympanic otitis, cholesteatoma, and tympanic membrane retractions. Otol Neurotol. 2012;33(6):934–40.
- 184. Relucenti M, Miglietta S, Bove G, Donfrancesco O, Battaglione E, Familiari P, et al. SEM BSE 3D image analysis of human incus bone affected by cholesteatoma ascribes to osteoclasts the bone erosion and VpSEM dEDX analysis reveals new bone formation. Scanning. 2020;2020:9371516.
- Takayanagi H. RANKL as the master regulator of osteoclast differentiation. J Bone Miner Metab. 2021;39(1):13–8.
- Kuczkowski J, Kobierska-Gulida G, Izycka-Swieszewska E, Potocka M, Mikaszewski B, Sierszen W. Molecular control of bone resorption in chronic otitis media with cholesteatoma. Otolaryngol Pol. 2010;64(4):219–24.
- Chen AP, Wang B, Zhong F, Song GZ, Song HF, Yu K, et al. Expression levels of receptor activator of nuclear factor-kappaB ligand and osteoprotegerin are associated with middle ear cholesteatoma risk. Acta Otolaryngol. 2015;135(7):655–66.
- Jeong JH, Park CW, Tae K, Lee SH, Shin DH, Kim KR, et al. Expression of RANKL and OPG in middle ear cholesteatoma tissue. Laryngoscope. 2006;116(7):1180–4.
- 189. Ma Y, Ye S. The expression of receptor activator nuclear factor-kappaB ligand and osteoprotegerin in cholesteatoma. Lin Chung Er Bi Yan Hou Tou Jing Wai Ke Za Zhi. 2008;22(7):293–5.
- Kuczkowski J, Sakowicz-Burkiewicz M, Izycka-Swieszewska E. Expression of the receptor activator for nuclear factor-kappaB ligand and osteoprotegerin in chronic otitis media. Am J Otolaryngol. 2010;31(6):404–9.
- Xie S, Pan Z, Yin T, Ren J, Liu W. Expression of PTHrP and RANKL in acquired middle ear cholesteatoma epithelium. Acta Otolaryngol. 2020;140(5):351–5.
- Koizumi H, Suzuki H, Ikezaki S, Ohbuchi T, Hashida K, Sakai A. Osteoclasts are not activated in middle ear cholesteatoma. J Bone Miner Metab. 2016;34(2):193–200.
- Hamzei M, Ventriglia G, Hagnia M, Antonopolous A, Bernal-Sprekelsen M, Dazert S, et al. Osteoclast stimulating and differentiating factors in human cholesteatoma. Laryngoscope. 2003;113(3):436–42.
- 194. Xia M, Ding S, Zhang H, Liu F, Yin H, Xu A. Expression and significance of nuclear factor-kappa B ligand and correlation factor in the tissue of

otitis media with cholesteatoma. Lin Chung Er Bi Yan Hou Tou Jing Wai Ke Za Zhi. 2007;21(7):315–7.

- 195. Pereira MF, Pires FR, Armada L, Ferreira DC, Carrouel F, Bourgeois D, et al. Expression of inflammatory markers RANK, MMP-9 and PTHrP in chronic apical periodontitis from people living with HIV undergoing antiretroviral therapy. J Clin Med. 2020;9(11):3611.
- Cheshire IM, Blight A, Ratcliffe WA, Proops DW, Heath DA. Production of parathyroid-hormone-related protein by cholesteatoma cells in culture. Lancet. 1991;338(8774):1041–3.
- 197. Martin TJ. Osteoblast-derived PTHrP is a physiological regulator of bone formation. J Clin Investig. 2005;115(9):2322–4.
- Haruyama T, Furukawa M, Kusunoki T, Onoda J, Ikeda K. Expression of IL-17 and its role in bone destruction in human middle ear cholesteatoma. ORL J Otorhinolaryngol Relat Spec. 2010;72(6):325–31.
- 199. Tsubaki M, Komai M, Itoh T, Imano M, Sakamoto K, Shimaoka H, et al. Nitrogen-containing bisphosphonates inhibit RANKL- and M-CSFinduced osteoclast formation through the inhibition of ERK1/2 and Akt activation. J Biomed Sci. 2014;21:10.
- Kang J, Choi YJ, Seo BY, Jo U, Park SI, Kim YH, et al. A Selective FGFR inhibitor AZD4547 suppresses RANKL/M-CSF/OPG-dependent ostoclastogenesis and breast cancer growth in the metastatic bone microenvironment. Sci Rep. 2019;9(1):8726.
- Yari Z, Tabibi H, Najafi I, Hedayati M, Movahedian M. Effects of isoflavones on bone turnover markers in peritoneal dialysis patients: a randomized controlled trial. Int Urol Nephrol. 2020;52(7):1367–76.
- 202. Li CH, Ma ZZ, Jian LL, Wang XY, Sun L, Liu XY, et al. Iguratimod inhibits osteoclastogenesis by modulating the RANKL and TNF-alpha signaling pathways. Int Immunopharmacol. 2021;90:107219.
- 203. Tsubaki M, Satou T, Itoh T, Imano M, Yanae M, Kato C, et al. Bisphosphonate- and statin-induced enhancement of OPG expression and inhibition of CD9, M-CSF, and RANKL expressions via inhibition of the Ras/ MEK/ERK pathway and activation of p38MAPK in mouse bone marrow stromal cell line ST2. Mol Cell Endocrinol. 2012;361(1–2):219–31.
- Stern PH, Alvares K. Antitumor agent cabozantinib decreases RANKL expression in osteoblastic cells and inhibits osteoclastogenesis and PTHrP-stimulated bone resorption. J Cell Biochem. 2014;115(11):2033–8.
- Tengesdal IW, Kitzenberg D, Li S, Nyuydzefe MS, Chen W, Weiss JM, et al. The selective ROCK2 inhibitor KD025 reduces IL-17 secretion in human peripheral blood mononuclear cells independent of IL-1 and IL-6. Eur J Immunol. 2018;48(10):1679–86.
- Zanin-Zhorov A, Weiss JM, Trzeciak A, Chen W, Zhang J, Nyuydzefe MS, et al. Cutting edge: selective oral ROCK2 inhibitor reduces clinical scores in patients with psoriasis vulgaris and normalizes skin pathology via concurrent regulation of IL-17 and IL-10. J Immunol. 2017;198(10):3809–14.
- 207. Reiersolmoen AC, Han J, Sundby E, Hoff BH. Identification of fused pyrimidines as interleukin 17 secretion inhibitors. Eur J Med Chem. 2018;155:562–78.
- Wei Y, Sun X, Hua M, Tan W, Wang F, Zhang M. Inhibitory effect of a novel antirheumatic drug T-614 on the IL-6-induced RANKL/OPG, IL-17, and MMP-3 expression in synovial fibroblasts from rheumatoid arthritis patients. Biomed Res Int. 2015;2015:214683.
- Miyasato M, Takeno S, Hirakawa K. Expression of RANKL and proliferation abilities of cultured human middle ear cholesteatoma epithelial cells. Hiroshima J Med Sci. 2013;62(1):1–6.
- 210. Nason R, Jung JY, Chole RA. Lipopolysaccharide-induced osteoclastogenesis from mononuclear precursors: a mechanism for osteolysis in chronic otitis. J Assoc Res Otolaryngol. 2009;10(2):151–60.
- 211. Takayanagi H. New developments in osteoimmunology. Nat Rev Rheumatol. 2012;8(11):684–9.
- Wu Y, Tang X, Shao W, Lu Y. Effect of CT manifestations of cholesteatoma on MMP-2, MMP-9 and IL-6 in the serum of patients. Exp Ther Med. 2019;17(6):4441–6.
- 213. Artono, Surarto B, Purnami N, Hutahaen F, Mahardhika MR. The Association of IL-1 alpha level and TNF alpha expressions on bone destruction in chronic suppurative otitis media and cholesteatoma. Indian J Otolaryngol Head Neck Surg. 2020;72(1):1–7.
- Sastry KV, Sharma SC, Mann SB, Ganguly NK, Panda NK. Aural cholesteatoma: role of tumor necrosis factor-alpha in bone destruction. Am J Otol. 1999;20(2):158–61.

- 215. Mun SH, Park PSU, Park-Min KH. The M-CSF receptor in osteoclasts and beyond. Exp Mol Med. 2020;52(8):1239–54.
- Lorenzo J. The many ways of osteoclast activation. J Clin Investig. 2017;127(7):2530–2.
- Chiang CY, Kyritsis G, Graves DT, Amar S. Interleukin-1 and tumor necrosis factor activities partially account for calvarial bone resorption induced by local injection of lipopolysaccharide. Infect Immun. 1999;67(8):4231–6.
- 218. Zahid A, Li B, Kombe AJK, Jin T, Tao J. Pharmacological inhibitors of the NLRP3 inflammasome. Front Immunol. 2019;10:2538.
- Juliana C, Fernandes-Alnemri T, Wu J, Datta P, Solorzano L, Yu JW, et al. Anti-inflammatory compounds parthenolide and Bay 11-7082 are direct inhibitors of the inflammasome. J Biol Chem. 2010;285(13):9792–802.
- Kong R, Sun L, Li H, Wang D. The role of NLRP3 inflammasome in the pathogenesis of rheumatic disease. Autoimmunity. 2022;55(1):1–7.
- 221. Kluck V, Jansen T, Janssen M, Comarniceanu A, Efde M, Tengesdal IW, et al. Dapansutrile, an oral selective NLRP3 inflammasome inhibitor, for treatment of gout flares: an open-label, dose-adaptive, proof-of-concept, phase 2a trial. Lancet Rheumatol. 2020;2(5):e270–80.
- Rogosnitzky M, Danks R, Kardash E. Therapeutic potential of tranilast, an anti-allergy drug, in proliferative disorders. Anticancer Res. 2012;32(7):2471–8.
- 223. Jung JH, Kim SE, Kim HJ, Park K, Song GG, Choi SJ. A comparative pilot study of oral diacerein and locally treated diacerein-loaded nanoparticles in a model of osteoarthritis. Int J Pharm. 2020;581:119249.
- 224. Kang S, Tanaka T, Narazaki M, Kishimoto T. Targeting interleukin-6 signaling in clinic. Immunity. 2019;50(4):1007–23.
- 225. Genovese MC, Fleischmann R, Combe B, Hall S, Rubbert-Roth A, Zhang Y, et al. Safety and efficacy of upadacitinib in patients with active rheumatoid arthritis refractory to biologic disease-modifying anti-rheumatic drugs (SELECT-BEYOND): a double-blind, randomised controlled phase 3 trial. Lancet. 2018;391(10139):2513–24.
- 226. Richez C, Truchetet ME. Evaluating filgotinib for the treatment of rheumatoid arthritis. Expert Opin Pharmacother. 2021;22(18):2435–44.
- 227. Markham A, Keam SJ. Peficitinib: first global approval. Drugs. 2019;79(8):887–91.
- Reilley MJ, McCoon P, Cook C, Lyne P, Kurzrock R, Kim Y, et al. STAT3 antisense oligonucleotide AZD9150 in a subset of patients with heavily pretreated lymphoma: results of a phase 1b trial. J Immunother Cancer. 2018;6(1):119.
- Melagraki G, Ntougkos E, Rinotas V, Papaneophytou C, Leonis G, Mavromoustakos T, et al. Cheminformatics-aided discovery of small-molecule Protein-Protein Interaction (PPI) dual inhibitors of Tumor Necrosis Factor (TNF) and Receptor Activator of NF-kappaB Ligand (RANKL). PLoS Comput Biol. 2017;13(4):e1005372.
- 230. Xie S, Li S, Tian J, Li F. Iguratimod as a new drug for rheumatoid arthritis: current landscape. Front Pharmacol. 2020;11:73.
- Vaynrub A, Healey JH, Tap W, Vaynrub M. Pexidartinib in the management of advanced tenosynovial giant cell tumor: focus on patient selection and special considerations. Onco Targets Ther. 2022;15:53–66.
- Hou GQ, Guo C, Song GH, Fang N, Fan WJ, Chen XD, et al. Lipopolysaccharide (LPS) promotes osteoclast differentiation and activation by enhancing the MAPK pathway and COX-2 expression in RAW264.7 cells. Int J Mol Med. 2013;32(2):503–10.
- Catli T, Bayazit Y, Yilmaz A, Menevse A, Gokdogan O, Goksu N, et al. Quantitative measurement of m-RNA levels to assess expression of cyclooxygenase-II, inducible nitric oxide synthase and 12-lipoxygenase genes in middle ear cholesteatoma. Eur Arch Otorhinolaryngol. 2014;271(6):1471–5.
- 234. Wang S, Xie L, Zhang Y, Xu P, Liu A. Expression of prostaglandin E2 receptors in acquired middle ear cholesteatoma. Clin Exp Otorhinolaryngol. 2018;11(1):17–22.
- 235. Fidahic M, Jelicic Kadic A, Radic M, Puljak L. Celecoxib for rheumatoid arthritis. Cochrane Database Syst Rev. 2017;6:CD012095.
- Feng X, Tian M, Zhang W, Mei H. Gastrointestinal safety of etoricoxib in osteoarthritis and rheumatoid arthritis: a meta-analysis. PLoS ONE. 2018;13(1):e0190798.
- 237. Curtis E, Fuggle N, Shaw S, Spooner L, Ntani G, Parsons C, et al. Safety of cyclooxygenase-2 inhibitors in osteoarthritis: outcomes of a systematic review and meta-analysis. Drugs Aging. 2019;36(Suppl 1):25–44.

- 238. Ruperto N, Nikishina I, Pachanov ED, Shachbazian Y, Prieur AM, Mouy R, et al. A randomized, double-blind clinical trial of two doses of meloxicam compared with naproxen in children with juvenile idiopathic arthritis: short- and long-term efficacy and safety results. Arthritis Rheum. 2005;52(2):563–72.
- 239. Morales DS, Penido Nde O, da Silva ID, Stavale JN, Guilherme A, Fukuda Y. Matrix metalloproteinase 2: an important genetic marker for cholesteatomas. Braz J Otorhinolaryngol. 2007;73(1):51–7.
- 240. Olszewska E, Matulka M, Mroczko B, Pryczynicz A, Kemona A, Szmitkowski M, et al. Diagnostic value of matrix metalloproteinase 9 and tissue inhibitor of matrix metalloproteinases 1 in cholesteatoma. Histol Histopathol. 2016;31(3):307–15.
- 241. Zhu Z, Hong Y, Wang Y, He G, Ye S. The significance of keratinocyte in hyperproliferation of middle ear cholesteatoma. Lin Chung Er Bi Yan Hou Tou Jing Wai Ke Za Zhi. 2016;30(2):139–43.
- Kaya I, Avci CB, Sahin FF, Ozates NP, Sezgin B, Kurt CC, et al. Evaluation of significant gene expression changes in congenital and acquired cholesteatoma. Mol Biol Rep. 2020;47(8):6127–33.
- 243. Kan T, Ueda H, Takahara T, Tsuchiya Y, Kishimoto M, Uchida Y, et al. Association of matrix metalloproteinase-2 mRNA expression with subtypes of pediatric cholesteatoma. Biomed Res Int. 2021;2021:6644897.
- 244. Juhasz A, Sziklai I, Rakosy Z, Ecsedi S, Adany R, Balazs M. Elevated level of tenascin and matrix metalloproteinase 9 correlates with the bone destruction capacity of cholesteatomas. Otol Neurotol. 2009;30(4):559–65.
- 245. Fields GB. Mechanisms of action of novel drugs targeting angiogenesispromoting matrix metalloproteinases. Front Immunol. 2019;10:1278.
- Kumar GB, Nair BG, Perry JJP, Martin DBC. Recent insights into natural product inhibitors of matrix metalloproteinases. Medchemcomm. 2019;10(12):2024–37.
- 247. Cohen-Inbar O, Zaaroor M. Glioblastoma multiforme targeted therapy: the Chlorotoxin story. J Clin Neurosci. 2016;33:52–8.
- Miyanaga S, Sakurai H, Saiki I, Onaka H, Igarashi Y. Anti-invasive and antiangiogenic activities of naturally occurring dibenzodiazepine BU-4664L and its derivatives. Bioorg Med Chem Lett. 2010;20(3):963–5.
- Song R, Qiao W, He J, Huang J, Luo Y, Yang T. Proteases and their modulators in cancer therapy: challenges and opportunities. J Med Chem. 2021;64(6):2851–77.
- 250. Sparano JA, Bernardo P, Stephenson P, Gradishar WJ, Ingle JN, Zucker S, et al. Randomized phase III trial of marimastat versus placebo in patients with metastatic breast cancer who have responding or stable disease after first-line chemotherapy: Eastern Cooperative Oncology Group trial E2196. J Clin Oncol. 2004;22(23):4683–90.
- 251. Fujioka R, Mochizuki N, Ikeda M, Sato A, Nomura S, Owada S, et al. Change in plasma lactate concentration during arctigenin administration in a phase I clinical trial in patients with gemcitabine-refractory pancreatic cancer. PLoS ONE. 2018;13(6):e0198219.
- Oh C, Kim HJ, Kim HM. Vitamin D maintains E-cadherin intercellular junctions by downregulating MMP-9 production in human gingival keratinocytes treated by TNF-alpha. J Periodontal Implant Sci. 2019;49(5):270–86.
- Kobayashi H, Asano K, Kanai K, Suzaki H. Suppressive activity of vitamin D3 on matrix metalloproteinase production from cholesteatoma keratinocytes in vitro. Mediators Inflamm. 2005;2005(4):210–5.
- Harrison SR, Li D, Jeffery LE, Raza K, Hewison M. Vitamin D, autoimmune disease and rheumatoid arthritis. Calcif Tissue Int. 2020;106(1):58–75.
- Alippe Y, Wang C, Ricci B, Xiao J, Qu C, Zou W, et al. Bone matrix components activate the NLRP3 inflammasome and promote osteoclast differentiation. Sci Rep. 2017;7(1):6630.
- Lambert C, Zappia J, Sanchez C, Florin A, Dubuc JE, Henrotin Y. The damage-associated molecular patterns (DAMPs) as potential targets to treat osteoarthritis: perspectives from a review of the literature. Front Med (Lausanne). 2020;7:607186.
- Abramson M, Moriyama H, Huang CC. Histology, pathogenesis, and treatment of cholesteatoma. Ann Otol Rhinol Laryngol Suppl. 1984;112:125–8.
- Yan SD, Huang CC. Tumor necrosis factor alpha in middle ear cholesteatoma and its effect on keratinocytes in vitro. Ann Otol Rhinol Laryngol. 1991;100(2):157–61.
- 259. Sauder DN, Stanulis-Praeger BM, Gilchrest BA. Autocrine growth stimulation of human keratinocytes by epidermal cell-derived

thymocyte-activating factor: implications for skin aging. Arch Dermatol Res. 1988;280(2):71–6.

- Liu W, Xie S, Chen X, Rao X, Ren H, Hu B, et al. Activation of the IL-6/JAK/ STAT3 signaling pathway in human middle ear cholesteatoma epithelium. Int J Clin Exp Pathol. 2014;7(2):709–15.
- Chung JW, Yoon TH. Different production of interleukin-1alpha, interleukin-1beta and interleukin-8 from cholesteatomatous and normal epithelium. Acta Otolaryngol. 1998;118(3):386–91.
- Zhang C, Chen M, Chi Z. Cytokine secretion and pyroptosis of cholesteatoma keratinocytes mediated by AIM2 inflammasomes in response to cytoplasmic DNA. Mol Med Rep. 2021. https://doi.org/10.3892/mmr. 2021.11983.
- Bujia J, Kim C, Ostos P, Sudhoff H, Kastenbauer E, Hultner L. Interleukin 1 (IL-1) and IL-1-receptor antagonist (IL-1-RA) in middle ear cholesteatoma: an analysis of protein production and biological activity. Eur Arch Otorhinolaryngol. 1996;253(4–5):252–5.
- Yamamoto-Fukuda T, Aoki D, Hishikawa Y, Kobayashi T, Takahashi H, Koji T. Possible involvement of keratinocyte growth factor and its receptor in enhanced epithelial-cell proliferation and acquired recurrence of middle-ear cholesteatoma. Lab Investig. 2003;83(1):123–36.
- Ishibashi T, Shinogami M, Kaga K, Fukaya T. Keratinocyte growth factor and receptor mRNA expression in cholesteatoma of the middle ear. Acta Otolaryngol. 1997;117(5):714–8.
- Yoshikawa M, Kojima H, Wada K, Tsukidate T, Okada N, Saito H, et al. Identification of specific gene expression profiles in fibroblasts derived from middle ear cholesteatoma. Arch Otolaryngol Head Neck Surg. 2006;132(7):734–42.
- 267. d'Alessandro F, Raffa S, Mure C, Kovacs D, Torrisi MR, Barbara M. Paracrine loops of keratinocyte stimulation in cholesteatoma tissue: an immunofluorescence, transmission electron microscopy, and molecular study. Otol Neurotol. 2010;31(7):1163–9.
- Kojima H, Matsuhisa A, Shiwa M, Kamide Y, Nakamura M, Ohno T, et al. Expression of messenger RNA for keratinocyte growth factor in human cholesteatoma. Arch Otolaryngol Head Neck Surg. 1996;122(2):157–60.
- 269. Jin BJ, Min HJ, Jeong JH, Park CW, Lee SH. Expression of EGFR and microvessel density in middle ear cholesteatoma. Clin Exp Otorhinolaryngol. 2011;4(2):67–71.
- Liu W, Yin T, Ren J, Li L, Xiao Z, Chen X, et al. Activation of the EGFR/Akt/ NF-kappaB/cyclinD1 survival signaling pathway in human cholesteatoma epithelium. Eur Arch Otorhinolaryngol. 2014;271(2):265–73.
- Yamamoto-Fukuda T, Akiyama N, Shibata Y, Takahashi H, Ikeda T, Kohno M, et al. KGFR as a possible therapeutic target in middle ear cholesteatoma. Acta Otolaryngol. 2014;134(11):1121–7.
- Yamamoto-Fukuda T, Akiyama N, Shibata Y, Takahashi H, Ikeda T, Koji T. In vivo over-expression of KGF mimic human middle ear cholesteatoma. Eur Arch Otorhinolaryngol. 2015;272(10):2689–96.
- 273. Yamamoto-Fukuda T, Akiyama N. Keratinocyte growth factor signaling promotes stem/progenitor cell proliferation under p63 expression during middle ear cholesteatoma formation. Curr Opin Otolaryngol Head Neck Surg. 2020;28(5):291–5.
- 274. Yamamoto-Fukuda T, Akiyama N, Takahashi M, Kojima H. Keratinocyte growth factor (KGF) modulates epidermal progenitor cell kinetics through activation of p63 in middle ear cholesteatoma. J Assoc Res Otolaryngol. 2018;19(3):223–41.
- Yamamoto-Fukuda T, Takahashi H, Terakado M, Hishikawa Y, Koji T. Expression of keratinocyte growth factor and its receptor in noncholesteatomatous and cholesteatomatous chronic otitis media. Otol Neurotol. 2010;31(5):745–51.
- Harabagiu OE, Cosgarea M, Mogoanta CA, Leucuta DC, Maniu AA. Keratinocyte growth factor and its receptor expression in chronic otitis media with and without cholesteatoma. Rom J Morphol Embryol. 2017;58(4):1333–8.
- Sartore-Bianchi A, Pizzutilo EG, Marrapese G, Tosi F, Cerea G, Siena S. Entrectinib for the treatment of metastatic NSCLC: safety and efficacy. Expert Rev Anticancer Ther. 2020;20(5):333–41.
- Farago AF, Demetri GD. Larotrectinib, a selective tropomyosin receptor kinase inhibitor for adult and pediatric tropomyosin receptor kinase fusion cancers. Future Oncol. 2020;16(9):417–25.
- Turkili S, Gorur K, Ismi O, Serinsoz Linke E, Vayisoglu Y, Ozcan C. Differential expression of Ki-67 and P27 in cholesteatoma compared to skin

tissue predicts the prognosis of adult acquired cholesteatoma. J Int Adv Otol. 2021;17(4):306–12.

- Kara A, Duman BO, Yazir Y, SinanYilmaz M, Halbutogullari ZSU, Demir D, et al. Evaluation of the effect of diclofenac sodium and 5-fluourasil in a 3D cholesteatoma cell culture model. Otol Neurotol. 2019;40(8):1018–25.
- Olszewska E, Chodynicki S, Chyczewski L. Role of angiogenesis in the pathogenesis of cholesteatoma in adults. Otolaryngol Pol. 2004;58(3):559–63.
- Sudhoff H, Dazert S, Gonzales AM, Borkowski G, Park SY, Baird A, et al. Angiogenesis and angiogenic growth factors in middle ear cholesteatoma. Am J Otol. 2000;21(6):793–8.
- van Cruijsen H, Giaccone G, Hoekman K. Epidermal growth factor receptor and angiogenesis: opportunities for combined anticancer strategies. Int J Cancer. 2005;117(6):883–8.
- 284. Dungo RT, Keating GM. Afatinib: first global approval. Drugs. 2013;73(13):1503–15.
- 285. Deeks ED. Neratinib: first global approval. Drugs. 2017;77(15):1695-704.
- 286. Shirley M. Dacomitinib: first global approval. Drugs. 2018;78(18):1947–53.
- Kazandjian D, Blumenthal GM, Yuan W, He K, Keegan P, Pazdur R. FDA approval of gefitinib for the treatment of patients with metastatic EGFR mutation-positive non-small cell lung cancer. Clin Cancer Res. 2016;22(6):1307–12.
- Faivre S, Demetri G, Sargent W, Raymond E. Molecular basis for sunitinib efficacy and future clinical development. Nat Rev Drug Discov. 2007;6(9):734–45.
- 289. Chen Y, Tortorici MA, Garrett M, Hee B, Klamerus KJ, Pithavala YK. Clinical pharmacology of axitinib. Clin Pharmacokinet. 2013;52(9):713–25.
- 290. Wind S, Schmid U, Freiwald M, Marzin K, Lotz R, Ebner T, et al. Clinical pharmacokinetics and pharmacodynamics of nintedanib. Clin Pharmacokinet. 2019;58(9):1131–47.
- 291. Hui R, Pearson A, Cortes J, Campbell C, Poirot C, Azim HA Jr, et al. Lucitanib for the treatment of HR(+)/HER2(-) metastatic breast cancer: results from the multicohort phase II FINESSE study. Clin Cancer Res. 2020;26(2):354–63.
- Milewski C, Fedorowski A, Stan AC, Walter GF. Basic fibroblast growth factor (b-FGF) in the perimatrix of cholesteatoma. HNO. 1998;46(9):804–8.
- 293. Huisman MA, de Heer E, Ten Dijke P, Grote JJ. Transforming growth factor beta and wound healing in human cholesteatoma. Laryngoscope. 2008;118(1):94–8.
- 294. Lang S, Schilling V, Wollenberg B, Mack B, Nerlich A. Localization of transforming growth factor-beta-expressing cells and comparison with major extracellular components in aural cholesteatoma. Ann Otol Rhinol Laryngol. 1997;106(8):669–73.
- 295. Hamed MA, Sayed RH, Shiogama K, Eltaher MA, Suzuki K, Nakata S. Localisation of basic fibroblast growth factor in cholesteatoma matrix: an immunochemical study. J Laryngol Otol. 2019;133(3):183–6.
- 296. Yang X, Li X, Ma M, Zhang L, Zhang Q, Wang J, et al. Expression of transforming growth factor-beta 1 matrix metalloproteinase-1 and its inhibitor in human middle ear cholesteatoma. Zhonghua Er Bi Yan Hou Ke Za Zhi. 2002;37(2):121–3.
- 297. Xie Y, Su N, Yang J, Tan Q, Huang S, Jin M, et al. FGF/FGFR signaling in health and disease. Signal Transduct Target Ther. 2020;5(1):181.
- Massaro F, Molica M, Breccia M. Ponatinib: a review of efficacy and safety. Curr Cancer Drug Targets. 2018;18(9):847–56.
- 299. Hidaka H, Izumi N, Aramaki T, Ikeda M, Inaba Y, Imanaka K, et al. Subgroup analysis of efficacy and safety of orantinib in combination with TACE in Japanese HCC patients in a randomized phase III trial (ORIENTAL). Med Oncol. 2019;36(6):52.
- Orbegoso C, Marquina G, George A, Banerjee S. The role of Cediranib in ovarian cancer. Expert Opin Pharmacother. 2017;18(15):1637–48.
- Gachpazan M, Kashani H, Hassanian SM, Khazaei M, Khorrami S, Ferns GA, et al. Therapeutic potential of targeting transforming growth factor-beta in colorectal cancer: rational and progress. Curr Pharm Des. 2019;25(38):4085–9.
- 302. Tran HC, Wan Z, Sheard MA, Sun J, Jackson JR, Malvar J, et al. TGFbetaR1 blockade with galunisertib (LY2157299) enhances anti-neuroblastoma activity of the anti-GD2 antibody dinutuximab (ch14.18) with natural killer cells. Clin Cancer Res. 2017;23(3):804–13.

- Kim CS, Lee CH, Chung JW, Kim CD. Interleukin-1 alpha, interleukin-1 beta and interleukin-8 gene expression in human aural cholesteatomas. Acta Otolaryngol. 1996;116(2):302–6.
- Fukudome S, Wang C, Hamajima Y, Ye S, Zheng Y, Narita N, et al. Regulation of the angiogenesis of acquired middle ear cholesteatomas by inhibitor of DNA binding transcription factor. JAMA Otolaryngol Head Neck Surg. 2013;139(3):273–8.
- Gluth MB, Kaufmann YC, Dornhoffer JL, Ferguson S. Immunotargeted photodynamic therapy for cholesteatoma: in vitro results with anti-EGFR-coated indocyanine green nanocapsules. Otol Neurotol. 2015;36(1):76–81.
- Wright CG, Bird LL, Meyerhoff WL. Effect of 5-fluorouracil in cholesteatoma development in an animal model. Am J Otolaryngol. 1991;12(3):133–8.
- 307. Sala DT. Topical applications of 5-fluorouracil in the medical treatment of cholesteatoma of the middle ear. Ear Nose Throat J. 1994;73(6):412–4.
- Takahashi H, Funabiki K, Hasebe S, Fukuda-Yamamoto T, Kaieda S, Iwanaga T, et al. Clinical efficacy of 5-fluorouracil (5-FU) topical cream for treatment of cholesteatoma. Auris Nasus Larynx. 2005;32(4):353–7.
- Szeto B, Chiang H, Valentini C, Yu M, Kysar JW, Lalwani AK. Inner ear delivery: challenges and opportunities. Laryngoscope Investig Otolaryngol. 2020;5(1):122–31.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

